

Global Industrial Energy Efficiency Benchmarking

An Energy Policy Tool

Working Paper

November 2010



Global Industrial Energy Efficiency Benchmarking

An Energy Policy Tool

Working Paper

November 2010



UNITED NATIONS INDUSTRIAL DEVELOPMENT ORGANIZATION

ACKNOWLEDGEMENTS

This working paper has been produced by the United Nations Industrial Development Organization (UNIDO) as a joint product of the Energy and Climate Change and Research Branches, under the guidance of Pradeep Monga, Director of the Energy and Climate Change Branch. Global Industrial Energy Efficiency Benchmarking: An Energy Policy Tool was written by Değer Saygın, (d.saygin@uu.nl), Martin K. Patel, (m.k.patel@uu.nl) from the Utrecht University, Group Science, Technology and Society/Copernicus Institute, Utrecht, Netherlands and Dolf J. Gielen (d.gielen@unido.org), Chief of the Industrial Energy Efficiency Unit, Energy and Climate Change Branch, UNIDO, Vienna, Austria. The paper was edited by Rob Wright, Director of Wrighthand Ltd.

The work has greatly benefited from the substantive guidance received from Ludovico Alcorta, Director of the Research Branch at UNIDO. Special thanks go to Morgan Bazilian, Olga Memedovich and Bettina Schreck (UNIDO) and to Ernst Worrell (Utrecht University) for their valuable comments.

Disclaimer

This document represents work in progress and is intended to generate comment and discussion. It is not a fully polished publication. The views expressed herein are those of the author(s) and do not necessarily reflect the views of UNIDO. The opinions, statistical data and estimates contained in the text are the responsibility of the authors and should not be considered as bearing the endorsement of UNIDO.

A number of figures presented in the report originate from data and figures published in publicly available presentations or reports. Data have been redrawn to a single format for reasons of consistent and comparable display throughout the text with an effort to minimise information loss. Although great care has been taken to maintain the accuracy of such material and all other information herein, UNIDO does not assume any responsibility for consequences which may arise from the use of the material.

This document may be freely quoted or reprinted but acknowledgement is requested.

ABBREVIATIONS

The following abbreviations and acronyms appear in this document.

APP	Asia Pacific Partnership
BAT	Best Available Technology
BF	Blast Furnace
BOF	Basic Oxygen Furnace
BPT	Best Practice Technology
CCS	Carbon Capture and Storage
CEC	Capital and Energy Costs
СНР	Combined Heat and Power
CSI	Cement Sustainability Initiative
DC	Developing Countries
DRI	Direct Reduced Iron
EAF	Electric Arc Furnace
EC	European Commission
EEI	Energy Efficiency Index
EFTA	European Free Trade Association
EIT	Economies In Transition
EJ	Exajoule
EU	European Union
EU-ETS	European Union Emission Trading Scheme
FAOSTAT	Statistics Department of the Food and Agriculture Organization
GEA	Global Energy Assessment
GHG	Greenhouse gas
GJ	Gigajoule
GNR	Getting the Numbers Right
HVC	High Value Chemicals
IAI	International Aluminium Institute
IC	Industrialized Countries
IEA	International Energy Agency

IFA	International Fertilizer Industry Association
IIASA	International Institute for Applied Systems Analysis
IPTS	Institute for Prospective Technology Studies
ISIC	International Standard Industrial Classification
ITMF	International Textile Manufacturers Federation
MENA	Middle East and North Africa
MER	Market Exchange Rate
MI	Methanol Institute
OE	Open-end Spinning
OECD	Organization for Economic Co-operation and Development
PJ	Petajoule
PSI	Plant Survey International
SA	South America
SEC	Spefic Energy Consumption
SME	Small and Medium Sized Enterprises
UNIDO	United Nations Industrial Development Organization
UNSD	United Nations Statistics Division
VSBK	Vertical Shaft Brick Kiln
WSA	World Steel Association

TABLE OF CONTENTS

List	of F	igures	ix
List	of T	ables	х
Exe	cutiv	e summary	xi
1.	Intro	oduction	1
2.	Gen	eral methodology and the data sources	7
	a.	Benchmarking industrial energy use	7
	b.	Industrial energy use scenarios until 2030	11
	c.	Production cost analysis	12
3.	Ove	rall results	13
4.	Sce	narios	20
5.	Pro	duction cost analysis	23
6.	Disc	cussion	27
7.	Con	clusions	29
8.	Refe	erences	31
Ann	ex		37
Deta	ailed	sector results	37
	a.	Petroleum refineries	37
	b.	Chemical and petrochemical sector	38
		High value chemicals (HVCs) production in steam crackers	38
		Ammonia	41
		Methanol	41
	c.	Non-ferrous metals	42
		Alumina production	43
		Primary aluminium production	43
		Copper smelting	44
		Slab zinc production	45

d.	Iron and steel	45
e.	Non-metallic minerals industry	47
	Cement	48
	Lime production	49
	Glass production	49
	Ceramics	51
f.	Pulp and paper sector	52
g.	Foundries	53
h.	Textiles	54
	Spinning	54
	Weaving	55
i.	Food and beverages	56
	Brewery	58

List of Figures

Figure 1:	Sectoral breakdown of total final industrial energy use in OECD and non-OECD countries, 2007	2
Figure 2:	Illustrative energy benchmark curve for the manufacturing industry	7
Figure 3:	Total final industrial energy use worldwide, 2007-2030	20
Figure 4:	Indexed (Western Europe=1) energy use and energy costs for steam crackers and ammonia production in selected countries	24
Figure 5:	Specific Energy Consumption (SEC) versus capital and energy costs (CEC)	25
Figure 6:	Solomon benchmark curve (2000) and the estimated benchmark curve for the refinery sector (2003)	37
Figure 7:	Solomon benchmark curves and the estimated benchmark curve for steam crackers benchmark, 2005	39
Figure 8:	IFA benchmark curves (2004 and 2007) and the estimated benchmark curve (2007) for ammonia industry	40
Figure 9:	First estimate of benchmark curve for the methanol industry, 2006	41
Figure 10:	IAI benchmark curve and estimated benchmark curve for alumina production, 2007	42
Figure 11:	IAI benchmark curve and estimated benchmark curve for primary aluminium production, 2007	43
Figure 12:	Compiled benchmark curve for copper smelters based on Brook Hunt	44
Figure 13:	Brook Hunt benchmark curve for zinc smelters, 2006	45
Figure 14:	Estimated benchmark curve for the iron and steel industry, 2005	46
Figure 15:	CSI benchmark curve and estimated benchmark curve for clinker production, 2007	47
Figure 16:	CSI benchmark curve and estimated benchmark curve for cement production (electricity for grinding only), 2007	48
Figure 17:	Energy requirements of 123 continuous container glass furnaces, 1999 (normalized for 50% cullet)	50
Figure 18:	Estimated benchmark curve for the pulp and paper industry, 2006	53
Figure 19:	Compiled brewery benchmark curve based on KWA, 2003	58

List of Tables

Table 1:	Overview of data sources (production data and SEC) and the methodologies applied	10
Table 2:	Overview of ranges for average energy use and energy benchmark data	14
Table 3:	Comparison of estimated short-term industrial energy savings in industrialised and developing countries, 2007	18
Table 4:	Energy use of the steam cracking process (in GJ/t HVC) and the improvement potentials	39
Table 5:	IFA benchmark survey results for 2004 and 2007 (lower heating value, in GJ/t NH3)	40
Table 6:	Specific energy consumption of lime kilns in selected countries	49
Table 7:	Overview of specific energy consumption in continuous glass furnaces (in kJ/kg of melted glass)	50
Table 8:	Specific energy consumption of different brick making technologies in selected countries (in GJ/t)	51
Table 9:	Theoretical energy saving potentials in the pulp and paper sector compared to BAT, 2006	53
Table 10:	Foundry benchmark electricity use (in kWh/t of melted product, furnace electricity consumption)	54
Table 11:	Specific energy consumption values of spinning technologies in different countries	55
Table 12:	Specific energy consumption values of weaving technologies in different countries	56
Table 13:	Selected products and their specific energy consumption (in GJ of final energy per tonne of output unless otherwise stated) of the food and beverage sector in OECD countries	57
Table 14:	Results of the 1 st , 2 nd and 3 rd brewery benchmark (years refer to the benchmark surveys conducted)	58

EXECUTIVE SUMMARY

Worldwide, manufacturing industry accounted for a total final energy use of 127 Exajoules (EJ) in 2007¹. This is equivalent to one third of the total final energy consumption of the global economy².

Developing countries and the economies in transition account for 60% of industry's total final energy consumption. Industry has significantly improved its energy efficiency in recent decades. But industry's total energy use continues to grow as a result of continuing large increases in the volume of production. Production is expected to continue to expand very substantially in the coming decades, particularly in developing countries. As a result, modest energy efficiency improvement rates will not be sufficient to stabilise or decrease the sector's energy demand in absolute terms. In order to make significant reductions, ambitious energy savings measures need to be implemented.

As a first step, the wider adoption of Best Practice Technologies (BPT) would enable significant reductions in energy use in the short term. In this study, we assess the energy saving potential that could be realised by the wide scale implementation of BPT in number of industry sectors. For each sector, we analyse the worldwide improvement potential, distinguishing industrialised countries from developing countries and the economies in transition.

Benchmarking the energy use of manufacturing industry

International benchmarks, based on the energy use of BPTs, are estimated for a total of 26 industrial processes, products and industry sectors. These processes include the energyintensive sectors such as the iron and steel, and chemical and petrochemical sectors, as well as number of light industries and small-scale sectors such as foundries and lime kilns. The total energy used by these processes represents approximately 60% of industry's current final energy use.

For a number of sectors, for example for steam crackers and aluminium smelters, *international benchmarks* are estimated from energy benchmark curves which are based on actual company data. For those sectors for which benchmark surveys do not exist, the report develops and compares energy indicators in different regions to provide an estimate for an international benchmark. Energy indicators are estimated based on literature data, i.e. production statistics and international energy statistics, and country-level comparisons are based either on an Energy Efficiency Index (EEI) or on an average of current levels of Specific Energy Consumption (SEC).

¹ Including petroleum feedstocks consumed for petrochemicals production.

² Including total non-energy used in industry, transformation, energy, transport and other sectors.

The analysis shows the existence of a worldwide potential to save 31 EJ a year if all the processes reviewed were to operate at the level of the international benchmark. Excluding feedstock use, this is equivalent to a worldwide improvement potential of 26%, comprising a 15% to 20% potential improvement in industrialised countries and a potential improvement of 30% to 35% in developing countries and economies in transition. The potential saving varies sector by sector. The percentage improvement potentials are less than the worldwide average for energy-intensive processes and sectors, although most light industry processes show higher improvement potentials.

Both Best Practice Technologies (BPT) and Best Available Technologies (BAT) offer potentials in the short- and medium-term

The future energy use of the worldwide manufacturing industry is projected to 2030 based on energy efficiency scenarios. Industry's total final energy use is projected to increase from 106 EJ3 in 2007 to 172 EJ in 2030 in a scenario which envisages no further improvements in efficiency or to 136 EJ in 2030 in a baseline scenario which envisages an improvement rate of 1% a year. If all industrial processes were to reach the BPT level by 2030, industry's total final energy use would be only slightly lower than the baseline at 128 EJ. Still further savings would be achieved by adopting Best Available Technologies (BAT)⁴. These are some 5% to 15% more efficient than BPT. With BAT, industrial energy use would amount to 114 EJ in 2030. Newer technologies which are not yet developed could potentially offer even higher improvement rates and therefore result in a more significant level of energy and CO₂ emissions reductions.

Regional differences in achieved levels of energy efficiency

The work underpinning this report has attempted to gain a better understanding of the differences in energy efficiency in different parts of the world. The assumption that low energy prices will lead to higher SEC and vice versa has been tested. It has been found to hold only for a few sectors, for example partly for steam cracking. Higher SEC is a result of factors other than energy prices only. For example, the high capital cost of new technologies is found possibly to be holding companies back from investing in more efficient and newer technologies, particularly where economic instability raises interest rates.

Next steps for benchmarking industry's energy use

This study demonstrates the value of benchmarking as a basis from which to estimate improvement potentials and to provide valuable information on industrial energy use. However, the data used are subject to a number of uncertainties and need further refinement.

The following steps are needed to maximise the potential of benchmarks and Energy Indicators as tools for measuring industry's energy use performance:

- Regional coverage of the benchmark surveys is incomplete. Data need to be collected for those sectors for which no information is currently available, particularly in developing countries and economies in transition.
- Benchmarking surveys need to be extended. They need to cover more processes in the

³ Excluding petroleum feedstocks used for petrochemicals production.

⁴ See Section 2 for the definitions of BPT and BAT as used in this paper. The definitions used in this paper, however, may not be consistent with the BAT definition stated in the European Union Directive 96/61/EC which concerns the Integrated Pollution Prevention and Control (IPPC) (EC, 2008).

energy-intensive sectors. And they need to secure better coverage of the most important processes in light industries and small-scale clusters, most of which are made up of small and medium size enterprises. Understanding the energy use of these smaller scale industries is particularly important since they have relatively large improvement potentials in percentage terms.

- Calibration of the benchmark curves is needed to support the refinement of the international benchmark data. For an objective comparison between countries, local conditions, for example reflecting regional differences such as the availability and quality of raw materials and feedstocks, need to be accounted for. Plants currently in operation do not have control over such conditions which may constrain their ability to achieve the level of efficiency envisaged by the international benchmarks.
- More insight needs to be gained in the sectors which have not yet been analysed or where the analysis was limited to a few SEC data points only. These sectors include food and beverages, machinery, transport equipment, metals processing, construction, and leather, many of which are also large energy consumers. Most of these sectors create important value added in developing countries.
- International energy statistics, which are the basis of the EEI approach, are subject to uncertainties. Closer collaboration is required between energy experts in companies and international statistics offices for improving the quality of international energy statistics.
- A dedicated effort is required in developing countries, where industry sectors need to be informed about the importance of efficient energy use and encouraged to implement

measures to enable more effective energy management and monitoring.

Key findings

- The bulk of industrial energy use is accounted for by the production of a relatively small number of energy intensive commodities. Chemicals and petrochemicals and the iron and steel sector account for approximately half of all industrial energy used worldwide. Other sectors that account for a significant share of industrial energy use are non-ferrous metals, non-metallic minerals and the pulp and paper sector.
- SEC differs significantly between countries and sectors as a result of differences in resource availability, energy prices, plant size, the age of capital stock, local factors, capital costs, awareness, opportunity costs and government policies.
- The benchmarking of the industry sector's energy use can provide valuable insights regarding energy efficiency potentials. Based on BPT data, global improvement potentials, and those for countries and regions are estimated. EEIs can supplement the benchmark surveys. They can also be used to support the estimation of improvement potentials for sectors where benchmark data are not available.
- Based on benchmark data, the current energy saving potential in manufacturing industry and petroleum refineries is estimated to be 31 EJ. This is equivalent to an energy efficiency improvement potential of approximately 26% of the industry's current total final industrial energy demand worldwide⁵. Around a quarter of the total energy saving potential (8 EJ to 9 EJ a year) is located in the industrialised countries; three-quarters of the saving

⁵ Excluding petroleum feedstocks used for petrochemicals production, but including the energy use of petroleum refineries.

potential (24 EJ to 25 EJ a year) is in developing countries and the economies in transition.

- Approximately two-thirds of the total savings potential is in the most energy-intensive industrial sectors although the energy efficiency potential is lower in percentage terms in these sectors than in the non-energy intensive sectors and light industries.
- Realising these potential energy savings would result in a reduction of 3% to 4% in the total costs of production. Worldwide, the total current energy cost savings potential in industry is estimated to be around USD 230 billion a year (excluding the cost of the investments required to upgrade current levels of technology to BPT). Industrialised countries have the potential to save around USD 65 billion in energy costs. Developing countries and economies in transition have the potential to save around USD 165 billion, i.e. more than 70% of the global potential cost savings. These savings are equivalent to 2% of current industrial value added worldwide.
- Achieving BPT by 2030 would result in manufacturing industry using 162 EJ of final energy. Excluding feedstocks, this would represent an improvement of 1.2% per year in

energy efficiency between 2007 and 2030. The total energy use would be 26% lower than it would be in the absence of any energy efficiency improvements.

- Implementing Best Available Technology (BAT) offers potential energy savings of up to 34% by 2030, equivalent to an energy efficiency improvement rate of 1.7% a year. Total final energy use with BAT is estimated to be 149 EJ in 2030.
- Excluding the cost of investment to upgrade existing technologies, the implementation of BPT is estimated to offer the potential to save USD 365 billion in energy costs in 2030. The implementation of BAT would offer the potential to save USD 495 billion in energy costs in 2030.
- The drivers of energy efficiency differ from country to country and from industry to industry. In some sectors, energy efficiency is partly driven by high energy prices. But in some other sectors, the high capital cost of investment in new and efficient plants is a major limitation on the rate of efficiency improvement in industry. This is particularly the case in countries where the economy is unstable, and where interest rates are high.

1. INTRODUCTION

Industrial energy use is dominated by developing countries and economies in transition, and this dominance will increase further in the coming decades

In 2007, the industry sector worldwide used approximately 127 exajoules (EJ) of final energy⁶, accounting for more than one-third of global final energy use. OECD countries (generally industrialised and high-income countries, abbreviated as "IC" in this report) accounted for approximately 51 EJ, i.e. around 40% of industrial final energy use worldwide (Figure 1). The remaining 76 EJ is consumed in non-OECD countries, the majority of which are developing countries and economies in transition (together abbreviated as "DC" in this report). In some manufacturing sectors such as iron and steel and non-metallic minerals industries, a larger share of energy is consumed in DCs than in ICs. In 2007, the industry spent more than USD 1 trillion on final energy (in market exchange rate (MER) terms) to produce a global value added of approximately USD 8 trillion (for 2007) (for ISIC: 15-37) (MER; World Bank, 2009) giving an average energy intensity of about 9%. Energy costs as a proportion of production costs vary significantly between different end-products, amounting to as much as 80% of ammonia production costs and between 1% and 10 % in yarn making and the machinery sector.

The bulk of industrial energy use is accounted for by the production of a small number of energy intensive commodities

Energy-intensive sectors, especially the chemical and petrochemical sector and the iron and steel sector, dominate industrial energy demand in both ICs and DCs. They account for approximately 50% of the total final industrial energy use **(Figure 1)**.

From 1971 to 2007, the final energy demand of manufacturing industry in DCs more than doubled, growing by an average of 3.2% a year. In ICs over the same period, it practically remained constant or increased only slightly (IEA, 2009a, b). Global industrial energy use is projected to double by 2050 in the absence of any new policy measures, and to increase by at least 50% by 2050 compared to today's levels even with the most ambitious emissions reduction policy changes (IEA, 2009c).

DCs dominate global industrial energy use for a number of reasons. First, they tend to move along a fairly traditional path of economic development, which proceeds from agriculture to industry and then to the service industries. This path is driven by consumer preferences, first for food and housing, and then as people become wealthier, later on for leisure and health care.

⁶ Including petroleum feedstocks for petrochemicals. Final energy use is derived from the total quantity of energy commodities (e.g. fuels, electricity, steam) delivered to consumers for their principal economic activity, excluding fuel conversion or transformation activities as defined elsewhere in the energy balances.





Note: Data includes feedstock use for petrochemicals, coke ovens and blast furnaces. It excludes petroleum refineries' energy use which is reported under the transformation and energy sector in IEA's energy statistics.

In IEA energy statistics, ideally data are provided (according to International Standard Industrial Classification (ISIC) of All Economic Activities, Rev.3.1; UNSD, 2010) for all manufacturing industry Divisions (i.e. 13-37), except for Divisions 23, 25, 33, 36 and 36. Division 23 is reported under the *own use* item of the transformation and energy sector. The other ISIC Divisions are reported under the *non-specific* item of the industry sector. However, some countries, particularly those outside the OECD, deviate from this reporting approach. In these countries, a share of the energy use or the entire energy demand of an industry sector, despite the availability of a specific item in IEA energy statistics, is reported to the *non-specific* item (see further detailed discussion in the main text).

Source: IEA, 2009a, b

Economic development and growth requires materials-intensive infrastructure and buildings, which in turn require massive amounts of cement, steel and other building materials. This pattern of development has, for example, been very clearly seen in China in recent decades.

Second, DCs account for 80% of the global population. As these countries reach industrialisation levels similar to those of the industrialised world this will inevitably mean much higher absolute levels of industrial energy use in the DCs compared to the ICs. Energy intensive industries are fairly evenly distributed around the world. This reflects the relative and often counterbalancing competitive advantages that result from specialisation, from countries with cheap energy sources attracting more energy intensive industry, and from locating industrial activity close to markets. As a result, large volumes of energy intensive materials are exported from DCs to ICs in the form of commodities and semi-finished as well as finished products, and it is expected that this activity will continue (Davis and Caldeira, 2010).

A number of factors influence differences in energy efficiency and energy intensity between countries

Energy intensities are expressed in terms of energy use per monetary unit, such as per unit of value added. The energy intensity of different end-products and sectors differs enormously. The energy intensity of similar products and products also differs very substantially between countries. A range of factors can play a role in these differences, depending on the product, sector, and country:

Access to resources For many energy intensive products such as steel, cement and aluminium, access to resources and the quality of raw materials and feedstock play a key role in the energy intensity of production. For example, the production of steel from steel recycling requires worldwide 8 gigajoules (GJ) per tonne of steel, less than half the 20.6 GI per tonne of steel that is needed for production from iron ore (Worrell et al., 2007). But the amount of scrap available for recycling is limited, and depends particularly on levels of past consumption years or even decades previously. In DCs where demand grows rapidly, the availability of scrap lags behind steel demand. As a consequence the share of steel production from virgin iron ore is much higher in DCs than in ICs, resulting in a higher level of average energy demand per tonne of steel produced. Similarly, Indian aluminium refineries are at a disadvantage with respect to energy use because local supplies of bauxite, from which the aluminium is produced, have a high share of calcium in their ore which requires larger amounts of process energy for conversion than other ores.

- Energy prices Where local energy supplies are relatively cheap, there is little incentive to industry to reduce its energy use. This effect is evident in major fossil-fuel producing countries such as Iran, Russia, Saudi Arabia and South Africa.
- Plant size and age of capital stock Older plants tend to be smaller. Smaller plants are generally less efficient. Older plants also tend to employ less efficient technologies. Investment in new plants and more efficient technologies is often not economic because the marginal production costs from existing capital stock are much smaller than they would be from new plant that was required to amortise its investment cost. For example, outdated, inefficient plants have remained in production in countries such as Russia because the capital stock invested in the times of the Former Soviet Union. which was in any case designed for much higher volumes than current production levels, has remained functional. Fortunately, the counterpart to this phenomenon is that the major growth in demand that is expected in DCs is likely to be met largely from new investment which will tend to be in new, more efficient, plant.
- Local factors Equipment import policies, local suppliers' strategies and limited available expertise can act as barriers for the uptake of more energy-efficient technologies. For example, many industrial plants in Russia base their operations on Russian-produced motors. These motors are relatively cheap, but they are not able to match the efficiency and quality of imported motors.
- Capital cost Energy efficient equipment tends to be more expensive. Many energy efficiency improvements rely on investment, the cost of which has to be recovered over time. High interest rates for capital tend to decrease investment in energy efficiency. For example

in the Ukraine, interest rates on local currency loans have reached a level of 40% in recent years. This is bound to be a significant disincentive to companies wishing to invest in energy saving measures.

- Awareness and opportunity cost In many economies, capital availability is limited, particularly in DCs. Management must choose how to use the limited capital either for capacity expansion or for investing in increasing energy and material efficiency. In many cases, expansion will be a priority since additional output will generate more revenue then efficiency improvements. Companies also often lack a good understanding of their energy use, their energy saving opportunities and the related economics. Government policies such as voluntary agreements, white certificates and energy efficiency tax incentives can help to increase awareness and encourage steps to improve energy management in companies. UNIDO is especially focusing on supporting the development of Energy Management Standards and benchmarking which can serve a similar purpose.
- Government policies National governments or transnational bodies such as European Commission (EC) design and apply energy and climate policies. Some of these policies directly or indirectly concern industrial energy use. Cap and trade schemes such as the European Union Emission Trading Scheme (EU-ETS), long term business-to-government commitments such as the voluntary benchmarking covenants used in the Netherlands, and fiscal instruments such as tax incentives and subsidy schemes for energy efficiency measures can all play a part in helping to drive improvements in energy efficiency and reductions in greenhouse gas (GHG) emissions. Industries in regions with relatively lax or ineffective policy environments may be less energy efficient.

Benchmark data can provide valuable insights regarding energy efficiency potentials

In recent years, in response to the need to establish national CO₂ reduction targets, substantial effort has been directed at the analysis of sectoral energy efficiency potentials at national level. The credibility of such efforts is questionable since data for individual plants or sectors in individual countries were often unavailable. Data collection, availability and coverage all need to be improved in many countries. Benchmarking and indicators analysis have an important part to play in this process.

Benchmarking is used to compare the performance of individual plants with the most energy efficient plant(s) on a sector-by-sector basis. Energy benchmarking is part of a much wider use of benchmarking as a management tool. The results of sectoral benchmark studies can be summarised in *benchmark curves* in which the energy use of individual plants is plotted as a dependent variable from the most efficient to the least efficient plant, either as function of cumulative production or of the number of plants. The information from benchmark curves can be used to assess the relative performance of individual plants. It can also, where sufficient specific information is available and the coverage of the benchmark curve is fairly comprehensive, be used to estimate the aggregate savings potential at the level of an individual country, a region, or worldwide.

A benchmark curve contains valuable information about best practice technologies (BPT), i.e. technologies that are energy efficient and already applied in practice. The most energy efficient plants in the benchmark curves are not, however, necessarily users of the most efficient technologies. They may, rather, be plants that benefit from exceptionally favourable feedstock quality or other non-technology-related factors. Detailed information on the reasons for the position of a plant on the curve cannot be obtained from the benchmark curve itself.

Only a relatively small number of plants are involved worldwide in the energy-intensive bulk materials industry. Only around 200 integrated steel plants, 200 steam cracking installations, 400 ammonia plants, 200 aluminium smelters, and 2 000 large cement kilns are in operation worldwide. Together they account for half of the global industrial energy use. Data on their performance can be acquired relatively easily. For other sectors, however, especially those dominated by small and medium enterprises (SMEs), the number of plants increases substantially. The development of benchmark curves for these sectors presents a more significant challenge.

For some industries and countries, benchmark curves are readily available (Section 2 and Annex). But data is often much less readily available, and often less reliable, for DCs than for ICs. Even where benchmark curves are publicly available, it is often impossible to identify individual plants based on the information given. Plant data are often confidential because of antitrust regulations and market sensitivities. It is not therefore possible to develop detailed efficiency investment programmes based on a benchmark curve because it remains unknown which plants exactly are the ones with the high savings potential. Information from additional sources is needed to complement benchmark curves if governments or other organisations are seeking to target investments in energy efficiency.

Benchmark data can be supplemented by efficiency indicator data

If energy use data cannot be identified at plant level, it is sometimes possible to quantify energy efficiency improvement potentials by comparing the average energy use within a country or region with the comparable best practice plant in the world. Average energy use can be derived from publicly available information such as energy statistics and production data. But the resulting energy efficiency indicators are generally less sensitive than benchmarking data.

Goal of this report

This report compiles a range of benchmark curves and indicators for energy intensive industries and products. The data presented cover approximately 55% of final manufacturing industry energy use including energy use in refineries. The analysis differentiates between ICs and DCs.

Based on the benchmarking data, the study provides:

- (i) an overview of the current technical energy saving potentials based on today's BPT. This information can be used to assess the global energy savings potential. The report does not, however, address the economics of these savings or the best means of enabling their delivery. Some of these savings will be realised through normal market pressures. Others may require governmental intervention through policy, fiscal or economic measures. Some may remain uneconomic for decades;
- (ii) a simplified economic analysis to assist a better understanding of the reasons why one country has a higher level of energy efficiency than another. This analysis covers energy costs and capital investment costs and the economic circumstances in selected countries for number of sectors; and
- (iii) projections through to 2030 which analyse the potential effect of implementing BPT and other best available energy saving technologies.

The following hypotheses will be tested in this report:

- The current average global energy saving potential from implementing BPT is around 15% - 35%, depending on sector and on location as between ICs and DCs.
- A number of rational explanations exist for the observed range. The relative energy efficiency of a plant or national sector does not necessarily correlate with its relative competitiveness.
- Energy audit and investment appraisal data suggest that many energy efficiency opportunities exist which offer savings of 10% to 20% in energy use and which have payback periods of less than two years.
- Policies can change the decision making framework and accelerate the uptake of energy efficiency improvements, thereby reducing the gap between current practice and the technical energy efficiency potential.
- Companies that benchmark their energy use and deploy energy management systems tend to achieve annual efficiency gains 1% to 2% higher than companies without such systems.
- In many countries, industrial energy is subsidised. This generates economic activity, but it also acts as a disincentive for energy efficiency improvements.
- Old plants tend to be less efficient than newer ones. Accelerating capital stock turnover can help to enhance efficiencies, but in many cases only at considerable cost.

- Industry will, with current policies and practices, achieve savings of 1% a year in process energy between now and 2030. This is treated as a baseline scenario.
 Implementing BPT can increase the efficiency improvement performance to 1.2% a year assuming no major structural changes.
- Based on today's energy prices and assuming no change in the fuel mix, the total value of the fuel saving would amount to USD 315 billion a year in 2030 in the baseline scenario, and USD 365 billion a year if BPT are implemented in the short-term⁷. Three quarters of these savings would accrue in DCs.
- Materials and product re-design, and the development of energy and materials management services, will help to reduce the average energy intensity of manufacturing significantly.

Section 2 explains the methodology and provides an overview of the data used. Section 3 presents the results of the analysis for each industry sector with a breakdown for the most important countries and regions. Section 4 outlines the results of the scenario analysis. Section 5 discusses the relationship of energy costs and capital costs to the energy efficiency of a number of industrial products in selected countries. Section 6 discusses the validity of the report's findings in the light of known uncertainties and lists a number of major shortcomings concerning data quality and data availability. The report concludes with a range of recommendations to governments and industry associations.

⁷ Throughout this report, energy cost savings refer exclusively to the fuel savings as a result of energy efficiency improvements. They do not account for the cost of the investments required to upgrade the current technology to the level of the international benchmark.

2. GENERAL METHODOLOGY AND THE DATA SOURCES

a. Benchmarking industrial energy use

A typical benchmark curve plots the efficiency of plants as a function of the total production

volume from all similar plants or as a function of the total number of plants that operate at that level of efficiency or worse **(Figure 2)**.



FIGURE 2:

Illustrative energy benchmark curve for the manufacturing industry

Note: SECs of the BAT, BPT, last decile and the least efficient plants according to this study are shown in the figure. Information on the *x* and *y*-axes has been indexed for simplicity. Normally the information would be plotted to show the specific energy consumption per unit of physical production against the cumulative production realised in the relevant year (in physical terms). The energy efficiency index for BPT is normalised to 1 for the 1st decile production share (*i.e.* the point on the *x*-axis equivalent to 0.1). More detailed explanations of the methodology are provided in the main text.

The most efficient plants are represented to the left and lower part of the curve, and the least efficient plants to the right and higher part of the curve. The shape of benchmark curves would vary for different sectors and regions. However, typically a few plants are very efficient and a few plants are very inefficient. This is generally represented by the steep slopes of the benchmark curve before the 1st decile and after the last decile respectively. Between these two deciles, benchmark curves tend to display a broadly linear relationship between energy efficiency and the share of cumulative production. This relationship can be used to support a rough assessment of the energy efficiency potential for an industrial process, which is defined as 50% of the difference between the efficiencies observed at the first and last deciles.

The most efficient plants in the benchmark curve are used to define the BPT. This report uses the 1st decile as the BPT and defines this as the *international benchmark* (Figure 2). Where possible, the analysis uses physical production levels to define the deciles. Where the lack of data makes such an approach inappropriate or unreliable, deciles are based on the number of plants.

Global benchmark curves are available for the following sectors: steam crackers (Solomon, 2005 in Leuckx 2008), clinker production (CSI, 2009a), petroleum refineries (Solomon, 2000 in Matthes et al., 2008) and ammonia production (IFA, 2009a, b). Benchmark curves are also available for the cement industry, as compiled by the Cement Sustainability Initiative (CSI, 2009a), and for the aluminium industry, as compiled by the International Aluminium Institute (IAI, 2009a, b). The accuracy of these curves often suffers from incomplete data particularly for fast-growing DCs such as China. Some of these data gaps can be filled through literature survey, although the information is scattered and its compatibility with the results of published benchmark studies is often questionable.

Plant benchmark data are complemented by two further types of analysis based on (i) the average current specific energy consumption (SEC) by world region or country, and (ii) the Energy Efficiency Index (EEI) as developed by Phylipsen *et al* (2002) and Neelis *et al* (2007a) for the Netherlands.

The SEC analysis uses the average current SEC at country or regional level depending on data availability⁸. If SEC data are not available, energy statistics provide the only basis for assessing energy efficiency. Energy statistics provide information on energy use at sectoral level, thereby including all production processes within that sector.

The EEI approach estimates the EEI of country *j* for sector *x* with *i* production processes as follows:

$$EEI_{j,x} = \frac{TFEU_{j,x}}{\sum_{i=1}^{n} P_{i,j} \times BPT_{i,x}}$$
(1)

where, TFEU is the actual energy use of sector x as reported in Energy Balances prepared by International Energy Agency (IEA) (in petajoules (PJ) per year), P is the production volume of product *i* in country *j* (in mega tonnes (Mt) per year), BPT is best practice technology energy use for the production of product *i* (in GJ per tonne of output) and *n* is the number of products to be aggregated. On this basis, a country is the most

⁸ If SEC data at country or regional levels are not available, estimates are made wherever possible based on published information on the relative energy use across different regions.

efficient worldwide when all its processes for a given sector have adopted BPT. In that case, the country or region has an EEI of 1.

On the basis of these approaches, the energy efficiency improvement potentials in sector x and in country or region j are determined as:

$$= 1 - \frac{International benchmark (BPT or SEC _{lowest,x})}{SEC_{l,x}}$$

or = 1 - $\frac{EEI _{lowest,x}}{EEI_{l,x}}$ (2)

Supplementary datasets are provided wherever possible with a higher level of detail for individual plants. The most up-to-date available data is used; in general this is the year 2007. Where data availability constraints require, "nameplate" energy efficiency plant data are used. These do not necessarily capture the variations in efficiency that result from daily operational practices, the frequency and quality of maintenance activities or the application of measures for debottlenecking and continuous improvement (including retrofitting) that are likely substantially to change energy efficiency. If these aspects were accounted for, the SEC of the most energy efficient plants would probably be lower than the benchmark curves show (i.e. these plants would be more efficient); and the SEC of the least energy efficient plants would probably by higher than shown (*i.e.* these plants would be less efficient). The slopes of the benchmark curves would therefore probably be steeper at the beginning and at the end.

For some developing countries, it has not been possible to apply either SEC or EEI methodologies to some sectors, primarily due to limitations in the availability of data on physical production, SEC or sector-specific total final energy use as given by international energy statistics. For these sectors, a comparison is provided of the current average SEC in ICs and in DCs. The international benchmark for estimating energy efficiency potentials is then set by the lowest achievable SEC that is identical with the BPT energy use.

This report analyses the energy use of 26 sectors, processes and products. In 10 cases, benchmark surveys are used as the principal methodology. Among these 10 cases, indicators are used to support the benchmark surveys in 8 cases. Indicators are used alone to estimate improvement potentials in only 4 cases. In 12 cases, energy efficiency potentials are determined by reference to the limited comparison of SEC values. Table 1 and Table 2 (below) identify the methodologies that have been used in individual cases using the following annotations:

- "B" for benchmark survey data,
- "I" for *i*ndicators: *i.e.* average current SEC or EEI data, and
- "*L*" for *l*imited SEC comparisons.

One international benchmark is provided for each product. The data is insufficient to support a deeper differentiation between types of raw material, feedstock or plant size. The analysis focuses on energy use only.

The products analysed (denoted as *i* in Equation 1) for each sector are chosen according to data availability and the structure of the sector. For example, numerous production processes are operated in the refinery sector, leading to a wide range of products. The product mix differs substantially across countries. The most important processes operated in the refinery sector are combined into a single EEI. In the aluminium sector, by contrast, the EEI for primary aluminium smelters or ammonia production is based on a single product.

TABLE 1:

Overview of data sources (production data and SEC) and the methodologies applied

Sectors, products and processes	DATA Production data	SOURCES SEC	Methodology applie	d
Petroleum refineries	OGJ, 2003	Worrell and Galitsky, 2004; 2005; Neelis <i>et al.,</i> 2005	EEI	1
Chemical and petroch	emical			
High value chemicals (steam crackers)	OGJ, 2008	Solomon, 2005 in Leuckx, 2008; Saygin <i>et al.,</i> 2009; Lvarious	Regional SEC	B & I
Ammonia	USGS, 2009a	International Fertilizer Industry Association (IFA); Saygin <i>et al.,</i> 2009 and various	Regional SEC+Literature data	B & I
Methanol	MI, 2009	Various	Regional SEC	B & I
Non-ferrous metals				
Alumina production	IAI, 2009; USGS, 2009b	IAI, 2009a,b and various	Regional SEC	B & I
Aluminium smelters	IAI, 2009; USGS, 2009c			
Copper smelters	Brook Hunt		-	В
Slab zinc	Brook Hunt		Regional SEC	1
Iron and steel	WSA, 2009	Worrell <i>et al.,</i> 2007	EEI	1
Non-metallic minerals				1
Cement ¹	USGS, 2009d	CSI, 2009a	Regional SEC	B & I
Lime	-	Various	Limited SEC comparison	L
Glass	Various		Literature data	B & I
Ceramic ²			Limited SEC comparison	В
Pulp and paper	FAOSTAT, 2009a	IEA, 2009c	-	1
Textile				I
Spinning	Various		Limited SEC comparison	L
Weaving	Various		Limited SEC comparison	L
Food and beverage	FAOSTAT, 2009b	Various	-	
Breweries	FAOSTAT, 2009b	KWA, 2004	-	В
Cheese	FAOSTAT, 2009b	Xu <i>et al.,</i> 2009	Limited SEC comparison	L
Fluid milk	Various	Xu and Flapper, 2009	Limited SEC comparison	L
Ferrous and non-ferrous foundries ³	Modern Casting, 2008	Various	Limited SEC comparison	L

¹ Clinker production and grinding process are analysed separately.

² The sector includes the separate analysis of brick making, tile making and sanitaryware products.

³ Foundries include the separate analysis of casting iron, steel, aluminium and copper.

Table 1 provides an overview of the data sources used, the methodologies adopted, and the type of benchmark value that results. Further details are provided in the Annex. The *international benchmark* values that result from this study are based purely on publicly available benchmark surveys and other open literature sources. They have not been reviewed and agreed upon by industry experts. A review process would be desirable in order to address a number of techno-economic aspects that are beyond the scope of this report, such as whether the BPTs could be extended more widely to the relevant sector than the present work assumes.

The final results of this study provide three sets of information for each process:

- Ranges for average energy use. These provide information on the current average energy use in various regions and an estimated global average. These averages are based either on benchmark surveys (B) or on indicators (I). For sectors and products where the analysis depends on limited SEC comparison (L) a global average is not estimated.
- Energy benchmark data.⁹ These provide further information on:
 - The most energy efficient plant. This is referred to as the Best Available Technology (BAT);
 - The international benchmark (*i.e.* the plant at the 1st decile, as described above);
 - The last decile plant (*i.e.* the most efficient plant in the last decile); and
 - the least energy efficient plant in the entire dataset.

• *Coverage of the sector:* This provides information on the production coverage. This is estimated by comparing the data collected in this paper with publicly available production statistics. These data refer to the production coverage of the benchmark curves prepared within this study.

The SEC data cover the direct energy used at plants for manufacturing a specific product. They do not include the energy required for mining or for the manufacturing of raw materials or the energy used in producing the energy consumed at the plant (e.g. primary energy used in power plant for electricity production or the energy used in extraction activities for producing, for example, naphtha or natural gas). The total process energy use is provided as a single value, in GJ per tonne of output, which includes any fuel, steam and electricity use or, where relevant, feedstock use. The data distinguish between different energy types only if an individual production process is based solely on a specific energy type (e.g. electricity use at primary aluminium smelter).

The study also uses the improvement potentials estimated for ICs and DCs to estimate the absolute saving potentials in each region, by multiplying the improvement potentials (in %) per sector by the actual energy use of that sector as reported in international energy statistics.¹⁰

b. Industrial energy use scenarios until 2030

Four energy efficiency scenarios have been developed in order to give a better understanding of possible developments between 2007 and 2030. These are:

⁹ In cases where data availability made it possible to apply benchmark survey (B) and indicators (I) methodologies simultaneously, the analysis gives priority to the benchmark survey for estimating the energy benchmark columns since data which originate from benchmark surveys provide information based on individual plants and are therefore more reliable. The indicators (I) data are then used as a supplementary dataset to determine the energy use in various regions.

¹⁰ If production statistics and the current average SEC data permit, bottom-up analysis is used to estimate the current energy use of a sector. For example, the current energy use of cement production is based on clinker production (heat use) and grinding (electricity use). This total is then deducted from the non-metallic minerals sector energy use provided in IEA energy statistics to estimate the current energy use of the sector's other products such as glass and ceramics.

- (i) Frozen efficiency: no additional energy efficiency savings are made, *i.e.* the current levels of energy efficiency are not improved upon.
- (ii) Baseline efficiency: energy efficiency improves at a rate of 1% a year.
- (iii) BPT scenario: all plants are operating at the current levels of BPT by 2030. This is equivalent to an energy efficiency improvement of 1.2% a year in the period 2007 to 2030.
- (iv) BAT scenario: all plant is operating at current levels of BAT by 2030. This is equivalent to an energy efficiency improvement of 1.7% a year in the period 2007 to 2030.

Both the BPT and BAT scenarios are based on technology already available commercially somewhere in the world. They do not take into account future technology developments.

All these scenarios assume levels of production growth based on the IEA's Low growth scenario (IEA, 2009c)¹¹. In this scenario, no production growth is estimated for ICs from 2007 to 2030. The average growth in China and economies in transition is estimated to be between 1% and 1.4% a year. India, other developing Asia, Middle East and Africa are assumed to grow at a much higher rate ranging between 2.5 and 4% a year. On average, physical production growth is estimated at 2.1% p.a. worldwide. The scenarios in the present analysis do not take into account the effect of any possible future structural changes, such as a larger production share of secondary steel or aluminium, or the increased production of higher value added products in DCs.

c. Production cost analysis

As a means of assessing whether energy costs appear to have an impact on levels of energy efficiency in different countries, this study has multiplied the energy use as plotted in energy benchmark curves (see Annex) by energy prices, producing a set of *energy cost* curves, discussed in Section 5.

Capital costs are another important factor influencing commercial decisions on the implementation of energy efficiency measures. To gain insight into the effect of the capital costs on energy efficiency investments, the study has collected data on initial investment costs, most of which refer to investments in ICs. After correcting for the circumstances in DCs by distinguishing between state-owned/local company investments as opposed to foreign direct investment, the capital and energy costs (CEC) are plotted graphically against SEC values. The outcome of this analysis is also discussed in Section 5.

[&]quot; The IEA study is limited to the energy-intensive sectors, namely chemical and petrochemical, iron and steel, aluminium, pulp and paper and cement. The present analysis applies the IEA growth rate projections, approximating cement to the entire nonmetallic minerals sector and the growth of all other industry sectors on the basis of the average growth rate of these five energy-intensive sectors. The growth of petroleum feedstock use for petrochemicals is based on the physical growth of the basic chemicals which are feedstock consumers, i.e. high value chemicals from steam cracking processes, ammonia and methanol production. Due to lack of availability of regional growth data for crude oil processing, the scenario analysis excludes the activities of petroleum refineries.

3. OVERALL RESULTS

Data has been collected and analysed on regional energy use in respect of 26 energy intensive industry processes and a number of light industry sectors. These are discussed in more detail in the Annex. The findings are summarised, with an indication of the data coverage for individual products and processes, in Table 2.

For the sectors analysed (which consume an aggregate of around 74 EJ/yr¹²), the total energy that could be saved by the implementation of BPT is 16.3 EJ/year (Table 3). Around a quarter of the total energy saving potential (4 EJ to 5 EJ/year) is located in the ICs. Most of the potential saving (11.5 EJ to 12.5 EJ/year) exists in DCs. Upgrading all processes of these sectors to the international benchmark (or the level of BPT) would save around 26% of the industrial current final energy use worldwide. In some energy intensive sectors, e.g. steam crackers and aluminium, some production processes or products have improvement potentials of around 10% to 20%. Light industries such as brick making or foundries typically have larger improvement potentials than the average.

While the average energy efficiency potential in ICs amounts to approximately 15%, the potential in DCs is around 30% on average. In some sectors, given the prevalence of small-sized plants equipped with old technology, it is as high as 40% - 50%. For some processes, such as aluminium smelting, pulp and paper and cement production, several DCs appear to be more energy efficient than the average IC. This may be explained by regional circumstances, such as the local availability of alternative fuels and blending agents in cement production, or by the fact that many DCs, as they have expanded production, have been able to adopt modern, more efficient technologies.

Achievable savings in petroleum refineries amount to 0.7 EJ a year in ICs and 2.9 EJ a year in DCs. This adds another 3.6 EJ of potential saving, resulting in potential total savings in the global industry including refineries of around 19.9 EJ. The total final energy saving potential in the industry and in refineries is more than 6% of the global final energy use (347 EJ).

¹² This includes the share of petroleum feedstocks consumed in the steam cracking process (for HVC production) and in the production of ammonia and methanol.

			Range	es for average	energy us	e in Er	iergy benchma	rk data		
Sectors (products and processes) (year data refers to)	Meth. (B/I/L)	Units	Selected ICs	Selected DCs (incl. EIT)	Global average	Best Available Technology (BAT)	Intl. Benchmark/ Lowest EEI	Last Decile Plant (or region)	Worst Plant (or region)	Coverage of the data (%)
Petroleum refineries (2003) ¹	1	EEI	0.7-0.8	1.3-3.8	1.25	1				90
Chemical and petrochemical High value chemicals ³ (2005) Ammonia ³ (2007) Methanol ^a (2006)	B & I B & I B & I B & I	GJ/t HVC GJ/t NH3 GJ/t MeOH	12.6-18.3 33.2-36.2 33.7-35.8	17.1-18.3 35.9-46.5 33.6-40.2	16.9 41 35.1	10.6 23.5 28.8	12.5 31.5 30	22.6 43 38.5	33.6 58 58	75 100 80
Non-ferrous metals Alumina production ⁵ (2007) Aluminium smelting ⁶ (2007)	B& I B& I	GJ/t alumina MWh/t primary	10.9-15.5 14.8-15.8	10.5-24.5 14.6-15	16 15.5	7.4 13.4	7.8 14.2	14.2 17.1	18.4 20.8	100 >95
Copper ⁷ Zinc ⁸ (2006)	B I	aluminium GJ/t copper GJ/t zinc	- 15.2-19.7	- 16.7-37.2	13.8 23.6	6.3 -	7.4 15.2	22.1 -	50.9 37.2	50 100
Iron and steel $(2005)^{9}$	-	EEI	1.16-1.4	1.4-2.2	1.45	1	1.16	,	2.2	100
Non-metallic minerals Clinker ¹⁰ (2007) Cement ¹⁰ (2007) Lime ¹¹ Glass ¹² (~2000s) Brick making ¹³ (~2000s)	В & I В & I L В & I L L	GJ/t clinker kWh/t cement GJ/t lime GJ/t melt MJ/kg fired brick	3.3-4.2 109-134 3.6-13 4-10 1.5-3	3.1-6.2 92-121 5-13 6.8-7.8 0.75-11	3.5 109 6.5 -	2.9 56 	3 88 3.2 3.6 VSBK: 0.75 Tunnel: 1.5	4.4 133 - -	6.6 144 - 8.7 -	100
Tiles ⁴⁴ Sanitaryware ⁴⁴	7 7	GJ/t tile GJ/t sanitaryware	1.9-7.3 4.2-11.3	3.1-8.3 4.4-20			1.9 4.2	,	,	

 TABLE 2:

 Overview of ranges for average energy use and energy benchmark data

Global Industrial Energy Efficiency Benchmarking: An Energy Policy Tool

			Range	es for average	energy use	e in En	iergy benchmai	rk data		
Sectors (products and processes) (year data refers to)	Meth. (B/l/L)	Units	Selected ICs	Selected DCs (incl. EIT)	Global average	Best Available Technology (BAT)	Intl. Benchmark/ Lowest EEI	Last Decile Plant (or region)	Worst Plant (or region)	Coverage of the data (%)
Pulp and paper ¹⁵	-	EEI (heat & electricity)	0.93-1.73	0.43-2.29	1.31	1				100
Textiles Spinning ¹⁶	Γ	GJ/t yarn	Ring yarn: 3.5-3.6 0E: 2.57	Ring yarn 3.5-3.6 Other:	ŗ	,	Ring yarn: 3.40 0E: 2.44	1		,
Weaving ¹⁷	Т	GJ/t woven cloth	11-65	0.5-7.5 5-43						
Food and beverages Brewery (2007) Cheese ¹⁸ Fluid milk ¹⁸	B L	MJ/hl GJ/t GJ/t fluid milk product	- 4.3-35.2 3.1-6.5		229 -		156 1.8 0.3		. ,	
Foundries* Cast iron	T	kWh/t iron melt	Cupola: 950 Electric:	780-850					I	I
Cast steel Cast aluminium	7	kWh/t steel melt kWh/t Al melt	525-715 Electric: 525-715 Fuel-fired:	735						
		1/////////////////////////////////////	600-1250 Electric: 440-590 Electric:	590	ı	ı	,	,	,	,
	L		400-1100	065			,			

- ¹ ICs: OECD countries. DCs: EIT/China. A benchmark energy use for the 1st decile cannot be estimated. The lowest estimated EEI, for OECD Europe, is reported. The average is weighted and is estimated based on the EEI and the crude oil capacity of each region.
- $^{\scriptscriptstyle 2}$ ICs: Japan & Korea and North America.
- ³ ICs: Europe and NA. DCs: MENA and China. Data includes feedstock use.
- ⁴ ICs: Europe and NA. DCs: South America and India. Data includes feedstock use.
- ⁵ ICs: NA and Europe (incl. Russia). DCs: South America and China.
- ⁶ ICs: Oceania and Europe (incl. Central Europe and EIT). DCs: Africa and Asia.
- ⁷ Data refers to copper smelters.
- ^a ICs: Western Europe and Japan. DCs: South America and China. Data refers to slab zinc production in zinc smelters.
- ⁹ ICs: Asia/Pacific and NA. DCs: China and CIS.
- ¹⁰ ICs: Pacific and NA. DCs: India and CIS. All SEC data originates from Getting the Numbers Right (GNR) database (CSI, 2009a). GNR database, a voluntary and an independently managed database, covers on average 31% of total global cement production. While for some regions the coverage is as high as 80 to 90%, e.g. North America, Central America and Europe. For other regions it is quite low, e.g. 20% for CIS and only 4% for China. Data is approximated assuming that it represents the energy use of all plants in a given region. However, an exception is made for China since it accounts for approximately half of the global cement production and GNR database refers only to a limited fraction of this value. The average SEC for the remainder of clinker and cement production in China (96%) is estimated based on China total average of 4.1 GJ of thermal energy per tonne of clinker (IEA, 2009c) and a 115 kWh grinding electricity use per tonne of cement (IEA, 2007).
- Waste heat recovery is a standard process applied in kilns in many countries. Typically it is used for drying raw materials; however, steam production (if potential buyers exist) or power generation is also possible. Net electricity production (after accounting for turbine and boiler operations) is rewarded. If a higher specific electricity production than 0.08 GJel per tonne of clinker is desired, modifications in kiln operation are necessary which would then lead to higher fuel demand (CSI, 2009b). These are accounted for under fuel use of clinker production as reported in GNR database.
- As opposed to heat use in kilns, which is expressed per tonne of clinker, specific electricity consumption is reported separately and expressed per tonne of cement. A significant share of electricity consumption is for grinding.
- ¹¹ ICs: Europe and Canada. DCs: China and Thailand.
- ¹² This is the aggregate of flat and container glass production. SEC data refers to per tonne of melt glass (at furnace). It is corrected for 50% cullet.
- ¹³ ICs: Europe (modern industrial brick kilns). DCs: Asian countries (small-medium size and very small kilns).
- Low-end of the average energy use in DCs refers to VSBK technology. Despite a low SEC value, the technology has limitations in firing bricks which are >15-20% hollow since at higher hollow rates breakage is observed. Furthermore the level of quality of bricks is lower than bricks fired in tunnel kilns. In spite of these drawbacks, given the suitability of this technology in DCs and its remarkably low SEC we refer to it as one of the benchmarks.
- ¹⁴ ICs: EU. DCs: India and China. SEC for ceramic tiles (wall and floor tiles) refers to firing process in kilns only. Total SEC of firing is determined based on number of firing steps (typically once, but for glazed products twice) and kiln type (roller hearth kilns versus less energy efficient tunnel kilns). In EU, the lower- and higher-ends of SEC data refer to once-fired roller hearth kilns and twice-fired tunnel kilns respectively. Process steps related to raw material preparation are less energy-intensive (less than total of 1 GJ/t). Lower end of SEC data for DCs refer to roller kiln wall tile production in India and the higher-end refers to average SEC for ceramic tiles production in China (no technology is specified).
- Firing step during sanitaryware production is more energy intensive. Lower- and higher-ends of SEC data for ICs refer to the most efficient and least efficient roller hearth kilns and conventional tunnel kilns respectively. DC data refer to tunnel kilns in India and kilns in Malaysia (no technology is specified). EU reference documents (IPTS/EC, 2007) report much higher values, up to 32 GJ per tonne for sanitaryware production (including other processes: casting, drying, glazing and other treatment). Average SEC data for China are equally high, at more than 30 GJ per tonne, but the system boundaries of the data are not clear, and therefore we do not report these values.
- ¹⁵ EEI values less than 1 point to serious problems in energy statistics. A country or a region can only reach the minimum achievable EEI of 1 if it applies the BAT in all its processes. So an EEI less than 1 is technically impossible. Given the data uncertainties, we do not report any energy benchmark values in these cases (except for BAT).
- ¹⁶ ICs: Italy, US, South Korea. DCs: China and India, and Thailand and Indonesia.
- ¹⁷ ICs: Germany. DCs: Thailand
- ¹⁸ ICs: Western Europe and North America. Data refers to the averages of lowest and highest recorded SEC values in individual cheese and fluid milk plants operated in North America and Europe. International benchmark data refer to the lowest recorded SEC data in these regions. We have no data for DCs.
- ¹⁹ ICs: EU and North America. DCs: an individual company in India. Data for ICs are given separately for cupola furnaces (based on coke), fuel-fired (natural gas) and electric furnaces (only electricity, no fuel). All data is expressed in kWh per tonne of molten metal (or melt). No international benchmark is given. The data excludes the material losses in foundries.

The industrial sectors not covered in detail by this study¹³ consume around 53 EJ/year, equivalent to 40% to 45% of total final industrial energy use. If these sectors are assumed to present similar potential energy savings, an additional saving of approximately 11.1 EJ/year would be achievable (*i.e.* a total energy saving potential of 31 EJ). This would suggest an energy efficiency improvement potential of approximately 26% for the world as a whole, 15% to 20% in ICs and around 35% in DCs. These estimates exclude feedstock demands where no saving potentials are estimated for. If feedstock energy use is included in the comparison, total energy efficiency improvement potential reduces to 22%.

Worldwide, the largest absolute potential savings are in the energy-intensive sectors, *i.e.* (from highest to lowest potential) petroleum refineries, iron and steel, non-ferrous metals, non-metallic minerals (mostly cement), chemical and petrochemicals, and pulp and paper. The total savings achievable in these sectors are 17.4 EJ/year, equivalent to 56% of the total global saving potential of 31 EJ/year. These sectors account together for a similar share of the total potential savings in ICs and DCs. The remaining energy savings potentials are in non-energy intensive or light industries. In these industries the relative savings potentials can be very high in percentage terms, and the savings potential amounts to 44% of the total industry potential.

IEA studies (IEA, 2007; 2009c) have assessed the potential technical energy saving that would

result if the energy-intensive sectors¹⁴ were to adopt BPT. These studies estimated energy saving potentials at process level as being 11.4 EJ to 16 EJ in 2004 and 14.2 EJ in 2006, equivalent to improvement potentials of 21% to 30% in 2004 and 25% in 2006, excluding feedstock use. The improvement potential estimated in this study for the same sectors is, at 23%, at a similar level¹⁵.

Including other industry sectors and petroleum refineries, the present study estimates an improvement potential of 26%. This is a few percentage points higher than the IEA studies, reflecting the higher savings potentials of petroleum refineries and of a number of light industry sectors and some SME clusters such as lime production, ceramics, textiles and food.

To achieve these potential savings, all plants would need to be upgraded to the level of the relevant international benchmark, i.e. BPT. Whether such upgrades will be implemented depends on the economic viability of the upgrade and on the energy and climate policy environment at country level. The investment costs of specific energy efficiency technologies vary widely, depending on their level of maturity. The willingness of companies to undertake investments depends on the payback period. In general, the higher the share of energy costs as a proportion of the total production costs (see the right hand column in Table 3), the more likely it is that investments in energy efficiency will be undertaken.

¹³ These sectors are (in order of detail level provided in IEA energy balances): machinery, transport, mining and quarrying, wood and wood products, construction and non-specified sectors. The last term is subject to particularly large uncertainties in energy statistics. This is especially the case for DCs. Its share is less than 10% for ICs while in DCs it accounts for as much as one quarter of the total final industrial energy demand according to energy statistics.

¹⁴ The chemical and petrochemical, iron and steel, cement, pulp and paper and aluminium sectors. Their energy use accounts for approximately 56% of the industry's total final energy consumption excluding feedstock use.

¹⁵ The energy saving potential is 13.9 EJ compared to a total final energy use of 59.7 EJ (excluding feedstock use). When feedstock use for the selected chemical and petrochemical processes is included, the total final energy use is estimated as 70.4 EJ, against which energy efficiency improvements could reduce demand by around 20%.

On average, energy costs constitute around 10% to 20% of industrial production costs. In energy intensive sectors such as the chemical industry, energy costs can constitute anywhere from 20% to as much as 50% of production costs. The energy improvement potentials identified in this report suggest that production costs could be

reduced on average by 1% to 15% in ICs and by 3% to 30% in DCs. For small-scale industries, particularly important in the developing world, the production cost reduction is around 3% to 4% on average, but as high as 20% in some cases, such as in brick making.

TABLE 3:

Comparison of estimated	l short-term	industrial	energy	savings ir	industrialised	and	developing
countries, 2007							

	lmj pot	provement tential (%)	To pote	tal savings ential (EJ/yr)	Share of energy costs (%)
Sectors and products	ICs	DCs (incl. EIT)	ICs	DCs (incl. EIT)	ICs DCs (incl. EIT)
Petroleum refineries	10-25	40-45	0.7	2.9	50-60
Chemical and petrochemical			0.5	1.8	
Steam cracking (excl. feedstock)	20-25	25-30	0.4	0.3	ro-8r
Ammonia	11	25	0.1	1.3	50-05
Methanol	9	14	0.0	0.1	
<u>Non-ferrous</u>			0.3	0.7	
Alumina production	35	50	0.1	0.5	30
Aluminium smelters	F-10	F	0.1	0.15	25-50
Other aluminium sec.	5-10	5	0.1	0.15	35-20
Copper smelters		45-50	0.0	0.1	-
Zinc	16	46	0.0	0.1	-
Iron and steel	10	30	0.7	5.4	10-30
Non-metallic minerals			0.8	2.0	
Cement	20	25	0.4	1.8	25-50
Lime					40
Glass	30-35	40	0.4	0.2	7-20
Ceramics					30-50
Pulp and paper	25	20	1.3	0.3	15-35
Textile					5-25
Spinning	10	20	0.1	0.3	F-1F
Weaving					C ₁ _C
Food and beverages	25	40	0.7	1.4	1-10
Total (excl. refineries)	10-15	25-30	4.4	11.8	
Other sectors	10-15	25-30	2.5	8.7	_
Total of all sectors (incl. refineries)	15	30-35	76	22.4	-
(excl. Feedstock)	15-20	30-35	7.0	<u>د</u> کرج	

Source: IEA, 2009a, b; own estimates

Note: As far as possible, energy costs are given as a share of total production costs (total of fixed costs and variable costs, including depreciation).

Manufacturing industry is currently estimated to spend around USD 1 trillion a year on energy, 45% of which is spent in ICs and 55% in DCs. Savings in energy costs from implementing BPT would amount to USD 65 billion in ICs and USD 165 billion in DCs. The DC potential saving is more than 70% of the total worldwide. The total potential savings do not take account of the cost of the investment required to implement BPT.

The potential savings represent approximately 2% of industry's current value added worldwide. This is significant, and should act as an incentive for more efficiency measures. But it appears that

these savings are not a major driver for investment, particularly given the risk related to volatile energy prices and other factors. The way in which energy costs and other cost parameters influence the attained levels of industrial energy efficiency across different regions is further explored in Section 5.

The adoption of BAT would result in even larger energy savings than the 26% of the total energy use of manufacturing industry and petroleum refineries that could be achieved by the adoption of BPT. Adopting BAT would result in a saving some 5% to 15% higher (UN Energy, 2009).

4. SCENARIOS

Industrial energy use grows from 106 EJ in 2007 to 172 EJ in 2030 in the frozen efficiency scenario (Figure 3). The baseline scenario assumes an efficiency improvement rate of 1%, resulting in a decrease in total energy use in 2030 to 136 EJ, 21% lower than in the frozen efficiency scenario. In the BPT scenario in which it is assumed that all industries converge to the current levels of BPT by 2030, total industrial energy use reduces to 128 EJ. This is 26% lower than in the frozen efficiency scenario. It represents, on average, an annual rate of improvement of 1.2%. Those countries/regions that perform already at the level of BPT are assumed to improve their energy use by 0.5% a year. DCs, which currently have higher SEC than average, would be expected to improve their energy efficiency by 1.5% to 3% a year until 2030. In the BAT scenario, in which all industries adopt BATs by 2030, total energy use is 34% lower than in the frozen efficiency scenario at 114 EJ. This represents a rate of energy efficiency improvement of 1.7% a year.¹⁶

FIGURE 3: Total final industrial energy use worldwide, 2007-2030



Note: Values exclude feedstock use

¹⁶ The scenarios assume no savings in feedstock use for petrochemicals. Worldwide feedstock demand will increase by 2.1% a year from 20.7 EJ in 2007 to 33.4 EJ in 2030.
Assuming no change in energy prices from today's levels, the energy bill of the global industry sector will increase from USD 1 trillion in 2007 to USD 1.75 trillion a year in 2030. According to IEA's energy price projections to 2030 (IEA, 2009d), changes in energy prices could lead to an increase of USD 250 billion a year in total energy costs, i.e. to a total of USD 2 trillion a year. The efficiency improvements implicit in the baseline scenario would reduce energy costs by around USD 300 billion a year in 2030.

Worldwide annual energy cost savings amount to USD 365 billion in 2030 in the BPT scenario and to USD 495 billion in the BAT scenario, assuming no change in the current fuel mix. In practice, most of the more efficient technologies require fuel switching from the current fuel mix to more efficiently combusted but more expensive options such as natural gas. Allowing for this, energy cost savings are estimated to be approximately USD 100 billion and USD 150 billion lower than these estimates respectively, resulting in a total saving in 2030 of USD 260 billion in the BPT scenario and USD 310 billion in the BAT scenario¹⁷.

Similar projections of industrial energy use are also made in studies prepared by International Institute for Applied Systems Analysis (IIASA) (Global Energy Assessment, GEA, *in preparation*) and IEA (2009c). In GEA's *supply* and *efficiency* scenarios, energy efficiency is projected to improve by 1.5% and 2% a year from 2005 to 2050.¹⁸ The rate of improvement in the GEA *supply* scenario is higher than that in the BPT scenario. In the GEA *efficiency* scenario, industry improves its energy performance at a faster rate than in either the BPT or BAT scenarios. These GEA scenarios allow for the potential impact of new technologies, enhanced material flows, process integration and other system options such as combined heat and power (CHP), in addition to the implementation of BPT and BAT. None of these wider process improvement possibilities is assumed to play a role in the BPT and BAT scenarios in this report.

In the IEA scenarios (2009c), industrial energy use including feedstocks is projected to increase from 122 EJ in 2006 to 183 EJ in 2030 the Low growth *Baseline*³⁹ scenario and to 169 EJ in the *BLUE Map* scenario. The Baseline, BPT and BAT scenario projections for 2030 in the present study, including feedstocks, are slightly lower than these estimates at 170 EJ, 162 EJ and 148 EJ respectively.

The IEA projections assume the implementation of system measures both to increase energy efficiency and to reduce carbon emissions. These include the wider adoption of CHP, fuel switching for fuel and feedstock use, increased recycling, and carbon capture and storage (CCS). Such measures may result in efficiency gains in addition to process improvements. But some of them, for example CCS, may increase energy demand in order to achieve CO₂ emissions reductions.

¹⁷ This assumes no changes in the prices of combustible renewable and waste products which are consumed as fuels in industry, and that the fuel mix of blast furnaces and coke ovens is unchanged. All petrochemical feedstocks are assumed gradually to switch to natural gas by 2030.

¹⁸ The reference year for GEA projections is 2005 in which industry consumed 115 EJ of energy. In the GEA study, energy use in 2050 in the *supply* and *efficiency* scenarios is 250 EJ and 200 EJ, respectively. This is equivalent to an energy use in 2030 of 175 EJ in the supply scenario and 155 EJ in the *efficiency*.

¹⁹ The Baseline scenario in the present study assumes a 1% p.a. energy efficiency improvement rate for all sectors, unlike the IEA Baseline scenario.

The projections underpinning this report are based on a different set of assumptions. In particular, they assume that it is technically and economically feasible for companies to implement BPT or BAT in all processes by 2030. On this basis, the scenarios project process energy efficiency improvements of 26% or 34% by implementing BPT or BAT. In practice, it is unlikely that all industrial processes will convert to BPT or BAT in this timescale. But the implementation of wider energy saving and emission reduction measures beyond the level of processes, may provide additional savings which could enable similar levels of energy efficiency saving overall still to be achieved.

5. PRODUCTION COST ANALYSIS

The analysis in Section 3 above has identified a range of energy efficiency improvement potentials. These vary sector by sector. The energy efficiency improvement potential also differs within sectors, region by region. This section explores the likely causes of such regional differences, in particular with a view to establishing the extent to which production cost parameters drive improvements in energy efficiency.

As shown in **Table 3**, energy costs account for approximately 10% - 20% of the total production cost of industry's physical output. The remaining 80% - 90% of costs are made up of components such as the cost of capital, labour, and materials. Each of these components is influenced by many factors.

For example, the capital costs of a depreciated plant are practically zero while, for a new plant, they could represent a very substantial share of production costs. Capital costs are also influenced by the local economic situation in the country where the plant is built, the agreed payback duration of the loan, the prevailing interest rate, the depreciation approach (linear or non-linear), and potentially also by the type of investor, whether it be a local company, a foreign investor, or the state, either directly or through state-owned banking mechanisms. Other factors such as government subsidies, or duties and taxation regimes if the technology is imported, can also reduce or increase capital costs. Material and energy costs are dependent on the technology deployed and are also influenced by the location of the plant and its size. The local availability of minerals or of natural energy sources can have a major impact in reducing the costs associated with these components. Government subsidies and the availability of long term contracts also help drive down costs. Other factors such as investment in research and development or operation and maintenance costs may play a role in driving costs up or down relative to other industrial manufacturers, but their contribution is generally low for industrial commodities.

Business decisions, especially in the energy intensive sectors, are influenced by very high or very low energy costs. In Japan, high energy prices have driven the development of innovative energy efficiency measures, the implementation of which has reduced SEC. By contrast, in regions where energy prices are low, such as the Middle East and Russia, where there is ready access to large quantities of cheap energy resources such as oil and natural gas, companies have had little incentive to reduce energy use.

If energy costs were a major determinant of energy use, energy cost curves would be expected to have a flatter, less profiled shape than the energy benchmark curves discussed in Section 3 and in the Annex. In regions with high energy costs, inefficient plant should be taken out of production either entirely, or to be replaced by more efficient plant, thus flattening the energy curve. In regions with low energy costs, there would be little incentive to invest in more efficient and generally more capital intensive plant, with a similar effect on the energy cost curve.

The analysis for steam cracking appears to confirm this hypothesis (left side of **Figure 4**). But the change in shape is so extreme in this case that the overall slope is inverted. Investment in the most energy-efficient steam crackers in Japan has resulted in energy savings which more than offset the disadvantage of local high energy prices. The opposite is true for the energy inefficient plants located in Saudi-Arabia, where extremely low energy prices more than offset the cost of excessively high levels of energy use.

By contrast, the data for ammonia production (right side of **Figure 4**) do not support the hypothesis. The shape of the curve is irregular,

with much lower than expected efficiency levels in India and Ukraine, where high energy costs prevail. This suggests that energy prices cannot be the only significant driver of energy efficiency.

Capital investment costs may also influence energy efficiency as energy-efficient technologies, which tend to be newer and more complex; generally require higher investment than less efficient technologies.

Figure 5 (below) compares the SEC of a number of sectors with their associated capital and energy costs. Average efficiency plant is represented by white dots. This includes both plant which is already depreciated, which tends to have lower costs, and relatively new plant where investment costs still need to be paid off. Average efficiency plant is typically state-owned or operated by local investors. Plant which is as efficient as the international benchmark, *i.e.* new

FIGURE 4:

Indexed (Western Europe=1) energy use and energy costs for steam crackers and ammonia production in selected countries



Sources for fuel prices: Steam crackers: McKinsey, 2008; Chemweek, 2007; Ammonia EFMA, 2000; PotashCorp, 2009

plant with the most efficient technology today, is represented by red squares²⁰.

Figure 3 shows that, in general, DCs have higher levels of SEC than ICs. This may reflect the fact

that investment costs in DCs are generally higher, thereby slowing down rates of investment in newer and more energy efficient technologies. This conclusion may be supported by the observation that the Ukraine has a high SEC

FIGURE 5:

Specific energy consumption (SEC) versus capital and energy costs (CEC)



Note: Energy prices refer to: steam cracking, 2007; ammonia, 2009; primary aluminium smelters, 2009; yarn making, 2003.

In the absence of any reliable initial investment costs for yarn making, the analysis is relatively limited. NA: North America, NAfr: North Africa, NE Asia: North East Asia, SAr: Saudi Arabia, WE: Western Europe.

Sources: Primary aluminium smelters: IAI, 2009a; CENEf, 2008; Adams, 2010. Yarn making: ITMF, 2003 in Koc and Kaplan, 2007. Investment costs are based on, steam cracking: Worrell et al, 2000; IPTS, 2003; ammonia: EFMA, 2000; Lako, 2009; aluminium smelters: Gielen and van Dril, 1997;

²⁰ International benchmark technology is primarily developed in ICs. When it is implemented in an IC, the investment is generally provided locally. In DCs, novel technology is often brought in through foreign direct investment. It is assumed that the investment for new capital will be granted by a local bank in the local currency of the country which receives the investment and that no additional taxes and duties are charged for importing the equipment. Such taxes and duties can increase investment costs.

despite high energy prices. Investment costs in Ukraine are comparatively high, particularly due to very high interest rates for loans borrowed in the local currency. By contrast, although many Russian ammonia plants are as inefficient as those in Ukraine, it has much lower energy prices and a relatively stable economy that leads to lower interest rates. This entails a lower cost burden when firms invest in switching to the international benchmark. This suggests that the combination of high energy prices and high capital costs may be responsible for the unfavourable position of the Ukraine in **Figure 4**.

China has the highest energy use per tonne of ammonia produced and has relatively low energy costs. But when capital costs are accounted for, production in China becomes more expensive than in a number of other countries, such as North America, which are more energy efficient. The fact that China accounts for one-third of all global ammonia production and is projected to increase its global production share in the coming decades (IEA, 2009c) shows that there are factors other than relative energy efficiency which enable Chinese companies to sustain their competitive position in the global market.

Similarly, although ring-yarn production in India has only slightly higher SEC than the USA, Italy or Brazil, India is the second largest yarn producer and one of the largest exporters of manufactured textile and apparel products²¹.

In general, as shown in **Figure 4** by the red squares and white dots for the Middle East,

China and Western Europe, it is difficult for new efficient plant which is amortising its investment costs to compete in a given region with older. relatively inefficient plant which has already depreciated its capital investment. The difference between the total costs of old and new plants is particularly large for DCs. In ICs, investment in new energy efficient plant is incentivised by higher energy prices and lower initial investment costs, primarily due to lower interest rates. As a result, DCs generally face a bigger challenge than ICs in moving to energy efficient technologies. Sometimes lower land and infrastructure costs, cheaper labour for construction and for local technical services, and lower equipment and material costs can help to reduce the otherwise relatively high cost of initial investment in DCs. But it is clear from the analysis in this study that low energy costs and low capital costs are not of themselves sufficient to trigger the investment needed to enable countries to become leading producers of bulk materials.

Energy inefficient industries in different countries can become or remain internationally competitive by balancing their higher energy use with lower costs for other production factors. A number of industry sectors, such as ceramics, textiles, leather, foundry and other processed metal products are dominated by SMEs in developing countries on this basis, for example by capitalising on lower labour costs, cheaper raw materials, economies of scale and lower profit margins. Further analysis could provide valuable insight into the relationship between the significance of these production factors.

²¹ Product quality is another issue. For low-value added products produced in mass quantities such as ethylene, the quality of the output originating from different regions may be similar. But for other products such as steel and yarn, global competitiveness is determined not only by low production costs but also by quality.

6. **DISCUSSION**

This study has provided for the first time energy benchmark curves for ICs and DCs for a range of energy-intensive industries and light industries. The analysis suffers, however, from number of uncertainties which need to be taken into account in drawing conclusions.

The quality of the analysis inevitably depends on the quality of the available data. For example, benchmark curves based on the EEI approach suffer from uncertainties in the data reported in energy statistics (IEA, 2009c). The improvement potentials estimated based on EEI data need to be studied in more detail, especially since the sectors analysed with this method (iron and steel, pulp and paper, and petroleum refineries) are large energy consumers. Cross-comparisons with other studies also demonstrate the need for further investigation: the refinery estimates for 2003 in the present study (with a high level of aggregation) suggest an EEI range between o.8 and 4.5 compared to the EII results of the benchmark survey at individual plant-level which range between 0.5 and 1.5 (Solomon, 2000 in Matthes et al., 2008).

Coverage of the data is also an issue. The more limited the coverage, such as in respect of the aluminium industry or the ammonia industry, the lower the confidence that can be place in the reliability of the benchmark curves. To reduce this uncertainty, the present study has attempted to fill data gaps in reported benchmark curves with literature data to increase the production coverage. However, the data used for complementing the benchmark curves is also subject to uncertainty, primarily because the system boundaries are not fully clear.

For light industries, first attempts have been made to provide SEC data on the basis of literature reviews. Some of these data may refer to very specific technologies or circumstances in an individual plant and may therefore not be representative for the entire sector in a country or region.

For approximately 40% to 45% of total final industrial energy use worldwide, our analysis provides only a first estimate of existing energy efficiency potentials. While a large share of energy use in these sectors (e.g. leather, metals processing, transport equipment, construction) occurs in DCs, some of these sectors, such as wood and machinery, are also important energy users in ICs. The literature provides little evidence on SEC values or on energy efficiency for these sectors. In addition, some important energy-intensive products and processes are also excluded from the analysis, including other basic and intermediate chemicals and polymers in the chemical and petrochemical sector, and the production of nonferrous metals other than aluminium.

IEA energy statistics do not provide the energy use of most DCs for individual sectors, except for the iron and steel, chemical and petrochemical and non-metallic mineral sectors. The energy use of all other sectors is reported under a nonspecified category where data is combined to a single value. Even in cases where a more detailed breakdown of sectoral energy use is provided, it is possible that part of the energy use of specific sectors is also reported under the non-specified category. On average, the nonspecified category accounts for 20% of total final industrial energy use in ICs (excluding feedstocks) and for more than 50% in respect of some DCs. This makes it impossible to conduct reliable detailed analysis. In addition, production and energy use data are possibly missing for most small plants in the informal sector in DCs, such as those involved in brick making in India.

The analysis in this report relies predominantly on detailed fuel use data, including steam and feedstocks. In the absence of sufficient data, it has not been possible to perform an in-depth analysis for electricity use. Exceptions are sectors where electricity use dominates the sector's energy use, *e.g.* primary aluminium production. In industry, on average, 65% of electricity demand is consumed by motors and drives such as pumps, compressors and fans (de Keulenaer, 2004). Sector electricity consumption varies widely from as low as 50% in the machinery and metal sector to 90% in the non-metallic minerals sector (de Almeida et al., 2003). To estimate the electricity savings potential at sectoral and regional level, energy statistics would need to be improved so as to report electricity use by sector and by demand category.

Given these data availability and data quality shortcomings, further work is necessary to extend the analysis. First, a thorough review of the energy data used in collaboration with industry associations and experts would help to improve the coverage and quality of the data reported. Second, the coverage of benchmark curves should be extended to assess the performance of individual sectors, particularly in respect of the less energy-intensive sectors that are composed primarily of SMEs. Third, to support the estimation of reliable SEC data for both ICs and DCs the reliability and consistency of production and energy statistics needs to be improved. Next steps should also include the determination of uncertainty ranges for the improvement potentials of each sector, quantified by reference to the uncertainties in SEC data and energy statistics.

The scenarios show that large potential process energy savings could be achieved by the implementation of BPT and BAT. Beyond the process level, even higher reductions could be achieved through the wider use of CHP²², the more effective integration of energy and material flows, and recycling. The potential energy savings of such options needs to be examined in more detail. It would also be very useful for further work to be undertaken to examine the likely impact of investment costs on the achievability of energy cost reduction potentials.

Further work is needed on the cost benchmark curves. The simplified approach based on energy and capital costs adopted in this study can help support an initial discussion on business decision-making. For a deeper analysis, the methodology needs to be extended to cover byproduct credits, the prices of raw materials and other utilities and to account for sectoral characteristics (*e.g.* production in integrated sites). Collaboration with the finance sector, industry associations and statistics offices would help to improve the quality of the analysis and to gaining better insight to the industrial investment and production decision making.

Such further work would enable a better understanding of the reasons why energy efficiency improvements are undertaken in some parts of the world but to a much lesser extent in others.

²² Only a share of the energy savings related to steam production in CHP has been assigned to the industrial sector. The remaining savings (related to the co-production of electricity) is attributed to the power generation sector.

7. CONCLUSIONS

This report has identified the global energy use and energy efficiency potentials of a range of energy-intensive and less energy-intensive manufacturing industries in a benchmarking analysis. The regional performance of industrialised countries (OECD countries) is differentiated from that of developing countries and economies in transition (non-OECD countries). The global manufacturing industry including petroleum refineries could save 31 EJ a year by implementing BPT. This is equivalent to a savings potential of 26% of current energy use overall or a saving of 15% - 20% in ICs and a saving of 30% - 35% in DCs.

Benchmark analysis is feasible for a number of energy-intensive sectors that produce bulk materials. The present analysis is constrained by data gaps and by low production coverage in some areas. The quality of available data, particularly SEC data, production statistics and energy statistics, is variable. The lack of monitoring systems in many DCs contributes further data uncertainty. The governments and national statistics offices of these countries need to be more active in this area in order to ensure that production and SEC data are consistent and to increase the quality of energy statistics. Collaboration with governments that already achieve better data collection can help improve statistical systems and performance in these countries.

Further assistance from industry associations and international organisations would also help in

developing and applying standardised methodologies for energy management and energy efficiency. This would help raise awareness, particularly in DCs, of the importance of energy efficiency and enable a better understanding of the competitive advantage that can be gained from implementing measures for reducing energy use.

There is considerable room also for DCs to adopt policies which will encourage practical outcomes such as the training of relevant company staff with a view to improving data measurement and providing information on the potential for energy efficiency savings. The wide range of UNIDO existing activities may provide a strong basis for joint international efforts to achieve substantial improvements in energy use and the delivery of the industrial energy efficiency potentials estimated in this study. It is important that these collaborative efforts are extended to small and medium-sized enterprises where some of the highest improvement potentials are likely to exist, especially in DCs.

Industrial energy and climate policies should ideally be based on energy and emission benchmark surveys based on real data measured at companies. Currently, some sectors (*e.g.* aluminium, cement) are active in developing methodologies and accurate data collection through sectoral partnerships, while others (*e.g.* the chemical and petrochemical sector) are lagging behind due to sector-specific issues. Worldwide, all industry sectors need to be active in such agreements in order to improve the accuracy of the data used in energy and climate policymaking.

This study suggests that there is considerable potential to achieve further energy efficiency savings. In the short term, further analysis is needed to verify and improve the benchmark data (based on extended benchmark surveys and energy indicators) reported in this study. Future research should be directed towards (i) developing strategies for realising the BAT potentials in each industry at the level of processes, (ii) improving the technologies beyond process level such as CHP, process integration and motor systems, and (iii) developing and applying novel and cost-effective technologies that could yield even higher savings than the currently available BPTs.

This study concludes that a joint international effort is required, first for increasing data availability and monitoring in developing countries, and second for harmonising data quality and consistency across all countries to enable more reliable estimates of energy efficiency improvement potentials to be made.

Corporate strategies motivated by market conditions will have important impacts on the rate of improvement in energy efficiency worldwide. Policy makers need to develop a clearer insight into the decision making processes that drive investments in energy efficiency in both ICs and DCs.

8. REFERENCES

- Adams, R. (2010). The strategic impact of changing industry markets on the aluminium industry. Presented to TMS 2010, Seattle, February 14-18, 2010.
- AIT-Asian Institute of Technology (2003). Small and Medium Scale Industries in Asia: Energy and Environment. Brick and Ceramics Sectors. Regional Energy Resources Information Center, Asian Institute of Technology. Pathumthani, Thailand.
- APP (2008), "Status Review of Industry Indicators for Energy Saving", presentation at the Fifth Task Force Meeting of the Asia Pacific Partnership, Busan, Korea.
- Banarjee, R (2006). Overview of energy efficiency for glass furnace. Organised by: Energy Systems
 Engineering, ITT Bombay & Petroleum
 Conservation Research Association (PCRA). April 3, 2006. Bombay, India.
- Beerkens, R.G.C., and J. van Limpt (2001), Energy Efficiency Benchmarking of Glass Furnaces, Paper presented at the 62nd Conference on Glass Problems. University of Illinois at Urbana-Champaign, Illinois, United States.
- BHP Billiton (2006). Hillside & Bayside aluminium.Presentation to Analysts. 29 March 2006.Melbourne, Australia.
- Chaisson, J (2009). Black carbon from brick kilns. Presented at Joint IIASA and University of Illinois Workshop. HarmonE: Today's technology, tomorrow's Drivers. May 19, 2008. Laxenburg, Austria.
- CENEf-Center for energy efficiency (2008). Bashamkov, I., K. Borisov, M. Dzedzichek, I. Gritsevich, A. Lunin. Resource of energy efficiency in Russia: Scale, costs and benefits. Developed for the World Bank. Moscow, Russia.

- Chemweek (2007). Chemical Week Price Report. Various issues in 2007. Chemical Week, New York, USA.
- CIEEDAC-Canadian Industrial Energy End-Use Data and Analysis Center (2004). Development of Energy Intensity Indicators for Canadian Industry 1990-2002. Burnaby, BC, Canada, http://oee.nrcan.gc.ca/publications/infosource/pub/c ipec/AnnualReporto2-03/annualReport-29.cfm?attr=20.
- COSMile (2010). Options to improve the performance: Brick clusters of Kolar and Malur, Karnataka. http://www.cosmile.org/doc/bricks_kolar.pdf. The Energy & Resources Institute, New Delhi, India.
- CRA-Corn Refineries Association (2010). World Corn Production, Consumption and Stocks. http://www.corn.org/wcornprod.htm.
- CSI-Cement Sustainability Initiative (2009a). Global Cement Database on CO2 and Energy Information. World Business Council for Sustainable Development. Geneva, Switzerland. http://www.csi.org
- CSI-Cement Sustainability Initiative (2009b). Development of State of the Art-Techniques in Cement Manufacturing: Trying to Look Ahead (CSI/ECRA-Technology Papers). European Cement Research Academy (ECRA). 4 June 2009. Düsseldorf/Geneva.
- Dankers, A.P.H. (1995). The energy situation in Thai lime industry. Energy for Sustainable Development 1, pp. 36-40.
- Davis, S.J. and K. Caldeira (2010). Consumption-based accounting of CO2 emissions. Proceedings of the National Academy of Sciences of the United States of America 107 (12), pp. 5687-5692.

- De Almedia, A.T., P. Fonseca, P. Bertoldi (2003). Energyefficient motor systems in the industrial and in the services sectors in the European Union: characterisation, potentials, barriers and policies. Energy 28, pp. 673-690.
- De Keulenaer, H. (2004). Energy Efficient Motor Driven Systems, Energy & Environment 15 (5): 873-905.
- Dhar, K., B. Gor, A. Nangia (2008). Excellence in Energy Management in 2008. Reliance Industries Ltd. Mumbai, India.
- EC-European Commission (2008). Directive 2008/1/EC of the European Parliament and of the Council of 15 January 2008 concerning integrated pollution prevention and control. Official Journal of the European Union L 24, 21/1/2008, pp. 8-29.
- Ecofys (2009). Blueprint for emission allowance allocation methodology in the EU Emission Trading Scheme (2013 - 2020). Utrecht, the Netherlands. http://www.ecofys.com/com/publications/reports_bo oks.asp.
- EFMA-European Fertilizer Manufacturers Association (2000). Best Available Techniques for Pollution Prevention and Control in the European Fertilizer Industry. Booklet No. 1 of 8: Production of Ammonia. Brussels, Belgium.
- EFMA-European Fertilizer Manufacturers Association (2008). Annual Report 2008. Brussels, Belgium.
- Energetics (1999). Energy and environmental profile of the U.S. metalcasting industry. Prepared for U.S. Department of Energy, Office of Industrial Technologies. September, 1999. Columbia, MD, USA.
- ESMAP-Energy Sector Management Assistance Program (2007). Bolivia: National Biomass Program. Report on Operational Activities. ESMAP Technical Paper 115/07. May 2007. Washington, DC, USA.
- EU (2009). Commission Directive 209/29/EC of the European Parliament and of the Council of 23 April 2009 amending Directive 2003/87/EC so as to improve and extend the greenhouse gas emission allowance trading scheme of the Community. Official Journal of the European Union. 5.6.2009
- EuLA-European Lime Association (2009). Developing a benchmark for the lime sector. 5 September 2009. Brussels, Belgium.
- FAO-Status and development issues of the brick industry in Asia (1993). Regional Wood Energy

Development Programme in Asia, FAO: GCP/RAS/154/NET. April 1993, Bangkok, Thailand.

- FAOSTAT-FAO Statistics (2009a). Production of pulp and paper products. Rome, Italy.
- FAOSTAT-FAO Statistics (2009b). Production of crops processed, live animals, livestock primary, livestock processed. Rome, Italy.
- Ferdausi, S.A., S. Vaideeswaran, S. Akbar (2008).
 Greening Brick Making Industries in Bangladesh.
 South Asia Social, Environment and Water
 Resources Management Unit. The World Bank. 12-14 November, 2008, Bangkok, Thailand.
- Gherzi (2009). Benchmarking of eleven (11) Sub-Saharan countries as a possible location for the setting up of a Cotton Yarn Spinning Mill. Project No: XP/RAF/08/005. Contract No: 16001581. Zürich, Switzerland.
- GEA-Global Energy Assessment. Prepared by International Institute for Applied Systems Analysis (IIASA). http://www.iiasa.ac.at/Research/ENE/GEA/index.htm, in preparation.
- Gielen, D.J. and A.W.N. van Dril (1997). The basic metal industry and its energy use. Prospects for the Dutch energy intensive industry. ECN-C-97-019, Petten, the Netherlands. March 1997.
- Gielen, D.J. (2009). Indicators and benchmarking: Issues and Recent Developments. Workshop International Energy Efficiency Standards to Promote Energy Efficiency and Reduce Carbon Emissions. Paris 17 March 2009. International Energy Agency. Paris, France.
- Gielen, D.J., T. Levissianos, S. Das (2009). Promoting energy efficiency and renewable energy in selected MSME clusters in India. GEF Project outline. 18 November 2009. New Delhi, India.
- Heierli, U. and S. Maithel (2008). Brick by brick: The Herculean task of cleaning up the Asian brick industry.
- IAI-International Aluminium Institute (2009a). Production and energy use statistics for primary aluminium and alumina production. London, United Kingdom.
- IAI-International Aluminium Institute (2009b). Aluminium Measuring & Benchmarking 2009. A report prepared for the Australian Government as

part of the Asia Pacific Partnership on Clean Development & Climate Aluminium Task Force. 3 August 2009. London, United Kingdom.

- IEA-International Energy Agency (2003a). Extended Energy Balances of non-OECD countries. Paris, France.
- IEA-International Energy Agency (2003b). Extended Energy Balances of OECD countries. Paris, France.
- IEA-International Energy Agency (2006a). Extended Energy Balances of non-OECD countries. Paris, France.
- IEA-International Energy Agency (2006b). Extended Energy Balances of OECD countries. Paris, France.
- IEA-International Energy Agency (2007). Tracking Industrial Energy Efficiency and CO2 Emissions. Paris, France.
- IEA-International Energy Agency (2009a). Extended Energy Balances of non-OECD countries. Paris, France.
- IEA-International Energy Agency (2009b). Extended Energy Balances of OECD countries. Paris, France.
- IEA-International Energy Agency (2009c). Energy technology transitions for industry: Strategies for the next industrial revolution. Paris, France.
- IEA-International Energy Agency (2009d). World Energy Outlook. Paris, France.
- IFA-International Fertilizers Association (2009a). Energy efficiency and CO2 Emissions in ammonia production. 2008-2009 Summary Report. September, 2009. Paris, France.
- IFA-International Fertilizers Association (2009b). Fertilizers, Climate Change and Enhancing Agricultural Productivity Sustainably. July, 2009. Paris, France..
- IPTS (2003). Reference document on Best Available Techniques in the Large Volume Organic Chemical Industry (2003). European Commission, Institute for Prospective Technological Studies. February, 2003. Seville, Spain.
- IPTS (2005). Reference document on Best Available Techniques in Smitheries and Foundries Industry (2005). European Commission, Institute for Prospective Technological Studies. May, 2005. Seville, Spain.

- IPTS (2006). Reference document on Best Available Techniques in the Food, Drink and Milk Industries. European Commission, Institute for Prospective Technological Studies. August, 2006. Seville, Spain.
- IPTS (2007). Reference document on Best Available Techniques in the Ceramic Industry. European Commission, Institute for Prospective Technological Studies. August, 2007.Seville, Spain.
- IPTS (2009draft). Reference document on Best Available Techniques in the Glass Manufacturing Industry. European Commission, Institute for Prospective Technological Studies. July, 2009draft. Seville, Spain.
- IPTS (2010). Reference document on Best Available Techniques in the Glass Cement, Lime and Magnesium Oxide Manufacturing Industries. European Commission, Institute for Prospective Technological Studies. May, 2010. Seville, Spain.
- Jokinen, J. Energy-efficiency developments in the North and South American and European Pulp & Paper Industry. IEA Workshop, October 9, 2006. Pöyry Forest Industry Consulting, Paris, France.
- Kobayashi, H., K.T. Wu, L.H. Switzer, S. Martinez, R. Giudici (2005). CO2 reduction from glass melting furnaces by oxy-fuel firing combined with batch cullet melting. Paper presented at XX. A.T.I.V. Conference Parma, Italy. September 14-16, 2005.
- Kirloskar (2009). Energy efficiency in Foundry-Kirloskarvadi Plant. September 2, 2009.
- Koc, E. and E. Kaplan (2007). An investigation on energy consumption in yarn production with special reference to ring spinning. Fibres & Textiles in Eastern Europe. October/December, 2007. Vol 15, No.4 (63), pp. 18-25.
- Kouw, J (2009). Nyrstar Budel Analyst Site Visit. 25 November 2009, Balen, Belgium.
- KWA Business Consultants & Brewing Research International (2004). Benchmarking energy efficiency worldwide in the beer industry 2003. 27 October, 2004. Feedback report to the participating breweries. Amersfoort, the Netherlands.
- Lako, P (2009). Energy conservation potential of the nitrogen fertilizer industry. ECN-E-09-011. March, 2009. Petten, the Netherlands.
- Lebbing, B. Energy use in Small-Scale Brick Making Industries in Northeast Thailand. Regional Wood

Energy Development Programme, Bangkok, Thailand, 1999.

Leuckx, D. (2008). An allocation method for petrochemicals. Presented at Ad hoc Group Energy, Feedstock & Logistics. Cefic, Brussels, Belgium, January 16th, 2008. http://ec.europa.eu/enterprise/sectors/chemicals/file s/wg_7_8fero8/14leuckx_sectoral_en.pdf.

- Liu, L. L. Aye, Z. Lu, P. Zhang (2006). Analysis of the Overall Energy Intensity of Alumina Refinery Process using Unit Process Energy Intensity and Product Ratio Method. Energy 31 (2006), pp. 1167-1176.
- Maithel, S (2002). Clean Technologies for improving Quality, Health and Safety in Small and Informal Sector. Paper for 2nd FICCI - TERI Global Conference "Green 2002: Agenda for Industry". February 7-8, 2002. New Delhi, India.
- Matthes, F.C., J. Repenning, E. Worrell, D. Phylipsne, N. Müller (2008). Pilot on Benchmarking in the EU ETS. Report prepared for the German Federal Ministry for the Environment, Nature Conservation and Nuclear Safety and the Dutch Ministry of Foreign Affairs. November 2008, Berlin/Utrecht.
- McKinsey (2008). Chemicals in the Middle East: Local Opportunities and Global Implications. December, 2008. Dubai, United Arab Emirates and Frankfurt, Germany.
- MI-Methanol Institute (2009). Global Methanol Capacity. Arlington, VA, USA. http://www.methanol.org/pdf/WorldMethanolPlants EndOf2006.pdf.
- Miller, M.M (2006). Lime 2005 Minerals Yearbook. US Geological Survey, June, 2006. Reston, VA, USA.
- Miner, R. and B. Upton (2002). Methods for estimating greenhouse gas emissions from lime kilns at Kraft pulp mills. Energy 27, pp. 729-738.
- Modern Casting (2008). 42nd Census of World Casting Production - 2007. December 2008. Schaumburg, IL, USA.
- Nagesha, N and M.H.B. Subramanya (2006). Energy efficiency for sustainable development of small industry clusters: What factors influence it. The International Journal of Economic Policy Studies. Volume 1, October 2006, Article 7.
- Nand, S. (2008). Recent efforts in energy conservation by fertiliser industry. New Delhi, India.

- Neelis, M., A. Ramirez, M. Patel (2004). Physical Indicators as a Basis for Estimating Energy Efficiency Developments in the Dutch Industry, Utrecht University, Utrecht, the Netherlands.
- Neelis, M., M.K. Patel., P.W. Bach, W.G. Haije (2005). Analysis of energy use and carbon losses in the chemical and refinery industries. ECN-1-05-008. September 2005. ECN (Energieonderzoek Centrum Nederland), Petten, the Netherlands.
- Neelis, M., A. Ramirez-Ramirez, M. Patel, J. Farla, P. Boonenkamp, K. Blok (2007a). Energy efficiency developments in the Dutch energy-intensive manufacturing industry, 1980-2003. Energy Policy 35, pp. 6112-6131.
- Neelis, M., M. Patel, K. Blok, W. Haije and P. Bach (2007b). Approximation of theoretical energy saving potentials for the petrochemical industry using energy balances for 68 key processes. Energy 32, pp. 1104-1123.
- NRCAN-Natural Resources Canada (2001). Energy efficiency opportunity guide in the lime industry -Canadian Lime Institute. Ottawa, ON, Canada.
- NRCAN-Natural Resources Canada (2003). Guide to energy efficiency opportunities in Canadian foundries. In partnership with the Canadian Foundry Association. Ottawa, ON, Canada.
- NRCAN-Natural Resources Canada (2005). Benchmarking energy use and costs in salt-anddry fish processing and lobster processing. Ottawa, ON, Canada.
- NRCAN-Natural Resources Canada (2008). Canadian ammonia producers: Benchmarking energy efficiency and carbon dioxide emissions. Ottawa, ON, Canada.
- Nurhayati, T., Y. Waridi and H. Roliadi. Progress in the technology of energy conversion from woody biomass in Indonesia. For. St. China 2006, 8 (3): 1-8.
- OGJ-Oil and Gas Journal (2003). World Refining Survey as of January 1st, 2004. December 22, 2003.
- OGJ-Oil and Gas Journal (2008). International survey of ethylene from steam crackers 2007.
- Oxenham, W. (2002). Current and future trends in yarn production. Journal of Textile and Apparel Technology and Management. Volume 2, Issue 2, Spring 2002.

- Partinen, J. (2008). Ferroally Technology Economical and Environmental Benefits. Outotec, Espoo, Finland.
- Phylipsen, D., K. Blok, E. Worrell, J. de Beer (2002). Benchmarking the energy efficiency of Dutch industry: an assessment of the expected effect on energy consumption and CO2 emissions. Energy Policy 30, pp. 663-679.

PotashCorp (2009). Nitrogen. Saskatoon, SK Canada.

- PREGA-Promotion of Renewable Energy, Energy Efficiency and Greenhouse gas Abatement (2005). Indonesia. Energy efficiency and energy conservation implementation at an integrated textile plean in Purwakarta. A pre-feasibility study report. October, 2005.
- Rajagopalan, S. Advances in weaving technology and looms.

http://www.pdexcil.org/news/40N1002/advances.ht.

- Ramirez, C.A. and K. Blok (2005). Energy use in the European Food Supply Chain. Interaction between Energy and Physical Flows. Ph.d thesis. Utrecht University. Utrecht, the Netherlands.
- Ramirez, C.A., M. Patel, K. Blok. From fluid milk to milk powder: Energy use and energy efficiency in the dairy industry. Energy 31 (12), pp. 1648-1668, 2006a.
- Ramirez, C.A., M. Patel, K. Blok. How much energy to process one pound of meat? A comparison of energy use and specific energy consumption in the meat industry of four European countries. Energy 31 (12), pp. 1711-1727, 2006b.
- Rauch, R. (2009). Indo-German Energy Program Opportunities for Cooperation. 3L Seminar Textile Industry. August 8th, 2009. Tirupur, India.
- Ray, B. K. and B.S. Reddy (2008). Understanding industrial energy use: Physical energy intensity chances in Indian manufacturing sector. WP-2008-011. Indira Gandhi Insitute of Development Research (IGIDR). Mumbai, India.
- Römpp (1995). Römpp Chemie Lexikon: J. Falbe and M. Regnitz, eds. Stuttgart, Germany.
- Rue, D.M., J. Servaites, W. Wolf (2007). Industrial Glass
 Bandwidth Survey. Final Report. Gas Technology
 Institute. Energy Utilization Center. August, 2007.
 Des Plaines, IL, USA.
- Sardeshpende, V., U.N. Gaitonde, R. Banerjee (2007). Model based energy benchmarking for glass

furnace. Energy Conversion and Management 48, pp. 2718-2738.

- Sarkisov, P.D., L.D. Konovalova, N. Yu. Mikhailenko, *E.G.* Vinokurov (2007). Glass Production in Russia
 Analysis and forecast. Glass and Ceramics. Vol.
 64, Nos. 11-12.
- Sathaye, J., L. Price, S, de la Rue du Can, D. Fridley (2005). Assessment of Energy Use and Energy Savings Potential in Selected Industrial Sectors in India. LBNL-57293. August 2005. Berkeley, CA, USA.
- Saygin, D., M.K. Patel, C. Tam, D.J. Gielen (2009). Chemical and Petrochemical sector: Potentials of best practice technology and other measures. IEA Information Paper. September 2009. Paris, France. www.iea.org/papers/2009/chemical_petrochemical_ sector.pdf.
- Schyns, V (2006). Towards a simple, robust and predictable EU Emission Trading Scheme. Benchmarks from concept to practice. March 21, 2006. http://www.sabic-europe.com/ourcommitments/environment/energy/_media/benchma rks_EU_ETS.pdf.
- Schwob, M.R.V., M. Henriques, A. Szklo (2009). Technical potential for developing natural gas use in the Brazilian red ceramic industry. Applied Energy 86, pp. 1524-1531.
- Sharpe, F. R., A.M. Pennartz, G. J. Freeman, G. Jackson (2009). Sustainable developments for the brewing industry. The Institute of Brewing & Distilling Africa Sec. -12th Scientific and Technical Convention.
- Spinnovation (2007). The Magazine for spinning mills. August 2007, No.23.
- SW-Sustainable Winegrowing. Improving energy use in the wine industry, Stage 1 project report.
- Turna, T (2002). Entegre tekstil prosesleri için uygun bilesik elektrik ve isi sistemlerinin belirlenmesi. VI. Ulusal Tesisat Mühendisligi Kongresi ve Sergisi. (in Turkish).
- Ullmann's (2007). Ullmann's Encyclopedia of Industrial Chemistry - the Ultimate Reference Release, 7th ed.
- UN Energy-United Nations Energy (2009). Policies and measures to realise industrial energy efficiency and mitigate climate change.
- UNSD-United Nations Statistics Division (2010). Detailed structure and explanatory notes ISIC Rev.3.1. http://unstats.un.org/unsd/cr/registry/regcst.asp?Cl=17.

- UNESCAP-United Nations Economic and Social Commission for Asia and the Pacific Colombo Plan Secretariat (2008). Unveiling Protectionism: Regional Responses to Remaining Barriers in the Textiles and Clothing Trade. New York, USA.
- UNFCCC-United Nations Framework Convention on Climate Change (2006). Clean Development Mechanism- Egyptian Brick Factory GHG Reduction Project. Submitted: December 22, 2006.
- USGS-U.S. Geological Survey (2009a). "Nitrogen" (Advance Release), Minerals Yearbook, United States Geological Survey, October, 2009. Reston, VA, 2009.
- USGS-U.S. Geological Survey (2009b). "Bauxite" (Advance Release), Minerals Yearbook, United States Geological Survey, October, 2009. Reston, VA, 2009.
- USGS-U.S. Geological Survey (2009c). "Aluminium" (Advance Release), Minerals Yearbook, United States Geological Survey, October, 2009. Reston, VA, 2009.
- USGS-U.S. Geological Survey (2009d). "Cement" (Advance Release), Minerals Yearbook, United States Geological Survey, October, 2009. Reston, VA, 2009.
- Venkatarama Reddy, B.V. and K.S. Jagadish (2003). Embodied energy of common and alternative building materials and technologies. Energy and Buildings, pp. 129-137.
- Visvanathan, C., S. Kumar, A. Priambodo, S. Vigneswaran (2000). Energy and environment indicators in the Thai textile industry.
- Wang, X (2007). China float glass energy efficiency survey. ITIBMIC. China.
- Warczok, A. and G. Riveros (2004). Energy efficiency in batch continuous and one-step copper pyrometallurgical processes. Presented at Improving energy efficiency in the APEC mining industry. October 21-23, 2004, Santiago, Chile. http://www.prien.cl/seminarios/taller_apec/en/index. htm.
- Wei, S (2007). Potentials and Strategies for Energysaving and CO₂ Emission Reduction in the Chinese Lime Industry. ITIBMIC. China.
- WorldBank (2009). World Development Indicators 2009. Washington, DC, USA. http://go.worldbank.org/UoFSM7AQ40.

- Worrell, E., D. Phylipsen, D. Einstein, N. Martin (2000).Energy use and the energy intensity of the U.S.chemical industry. LBNL-44314. Berkeley, CA, USA.April, 2000.
- Worrell, E. and C. Galitsky (2004). Profile of the Petroleum Refining Industry in California. LBNL-55450. March 2004. Berkeley, CA, USA.
- Worrell, E. and C. Galitsky (2005). Energy efficiency improvements and cost saving opportunities for petroleum refineries. An ENERGY STAR® guide for energy and plant managers. LBNL-56183. February, 2005. Berkeley, CA, USA.
- Worrell, E., L. Price, M. Neelis, C. Galitsky, Z. Nan (2007). World Best Practice Energy Intensity Values for Selected Industrial Sectors. LBNL-62806. June, 2007. Berkeley, CA, USA.
- WSA-World Steel Association (2009). Steel Statistical Yearbook 2008. Brussels, Belgium.
- Xiao-wu, W., G. Li-rong, Hua-Ben (2008). Resources conservation-The alternative scenarios for Chinese aluminium industry. Resources, Conservation and Recycling 52, pp. 1216-1220.
- Xu, T., J. Flapper, K. J. Kramer (2009). Characterization of energy use and performance of global cheese processing. Energy 34, pp. 1993-2000.
- Xu. T. and J. Flapper (2009). Energy use and implications for efficiency strategies in global fluid-milk processing industry. Energy Policy 37, pp. 5334-5341.
- Yanjia, W. and W. Chandler (2010). The Chinese nonferrous metals industry - energy use and CO2 emissions. Energy Policy 38 (2010), pp. 6475-6484.
- Zaiyin, Y (2007). Development Prediction about the Brick Industry by 2050. ITIBMIC. China.

Zunhua, T. and L. Xuemin (2008). Energy conservation and the development of China's electrolysis aluminium industry. Presented at IEA workshop. Paris, France. http://www.iea.org/work/2008/comparison/ KANG_Aluminum_EN.pdf.

Zwiers, R. (2009). Europees Klimaatbeleid en Emissiehandel: Uitdagingen voor de Kunstmestindustrie. MMF Thema Dag, 11 November 2009. Zeist, the Netherlands. (in Dutch)

ANNEX

Detailed sector results

This annex presents the benchmark curves prepared for individual industry sectors. It provides background information on sectors and technologies only where such detail is relevant to the understanding of the international benchmark data.

a. Petroleum refineries

In refineries, crude oil is processed into a wide range of refinery products. In all refineries, the first step is to separate crude oil into various fractions by means of atmospheric distillation. These fractions are then upgraded and blended to produce different oil-derived products.

Refineries convert crude oil to fuels and several **FIGURE 6:**

other products such as aromatics, lubricants, sulphur and many other compounds.

For the refinery sector, it is not possible to derive a meaningful single average SEC value for different world regions. Each of the numerous processes which take place in refineries has its own BPT value. The structure and the product mix of refineries vary within the same country and across the world. The analysis takes these differences into account by estimating an EEI for each country based on the BPT of 13 refinery processes²³ and the refinery structure of each country. The actual energy use of the sector is the sum of the final energy use and transformation losses as reported in IEA Energy Statistics (IEA, 2003a, b).





²³ These processes are atmospheric distillation, vacuum distillation, coking, thermal operations, catalytic cracking, catalytic reforming, catalytic hydrocracking, catalytic hydrotreating, alkylation, aromatics, lubricants, and the production of hydrogen and sulphur. These processes in 2000 accounted for 70% of the total final energy use of the global refinery sector (Neelis *et al.*,2005).

Note: the black line shows the regional EEI with respect to cumulative crude oil charge capacity (*primary x-axis*); the blue line shows the EII® of individual plants (*secondary* y-axis and *secondary* x-axis) denoted by study participation (expressed in %)).

Solomon Associates is an independent consulting firm that performs sectoral and product energy benchmarks, including for the refinery sector. The sector results prepared by Solomon Associates are expressed in an EII® (Energy Intensity Index) which accounts for the scale, location and complexity of refineries. A more detailed explanation of their approach can be found in Matthes *et al* (2008). Ell=100 is defined as standard energy use. A refinery with an EII below standard is more efficient than a refinery with EII higher than the standard. In **Figure 4**, in order to be comparable with the scale of EEI, we show Ell=1 as standard energy use instead of the typical Solomon approach where Ell=100.

References to study participation here and later in this report refer to the proportion of the total volume of production that is covered in the relevant benchmark survey. The coverage of individual surveys is in some cases significantly less than the total global production of the relevant sector. Where quoted, participation rates describe the percentage of overall global production that is covered by each survey.

Source: OGJ, 2003; IEA, 2003a, b; Solomon 2000 in Matthes et al., 2008; own estimates

According to this analysis, OECD Pacific, OECD Europe and OECD North America have EEI less than 1²⁴. This implies that they are more efficient on average than BPT. This is not technically possible. Other regions have EEIs as high as 4.5. These results point to limitations in the methodology and the data used. A country-level analysis is not possible for sectors such as the refinery sector where cogeneration and energy integration of the processes have significant impacts on levels of energy efficiency (Saygin *et al* (2009)). Site specific data are required for a meaningful analysis, as described in the main text of this report.

b. Chemical and petrochemical sector

Production of ethylene and other high value chemicals (HVC) such as propylene, butadiene (C4 fraction), benzene (aromatics) and hydrogen in steam crackers, together with the production of ammonia and methanol, account for more than half of the sector's total final energy use (including the related quantities of petroleum feedstocks). Some 95% of the sector's total final energy use is accounted for by approximately 60 processes (Neelis *et al.*, 2007b; Saygin et al., 2009). However, given the lack of publicly available data it is impossible to prepare benchmark curves for all these chemical rocesses.

High value chemicals (HVCs) production in steam crackers

Steam cracking is by far the largest energy user in the chemical and petrochemical sector, accounting for more than one third of the sector's final energy use including feedstocks (IEA, 2009c). Figure 7 shows the results of a Solomon Associates survey covering more than half of the global ethylene production capacity (Solomon, 2005 in Leuckx, 2008).

²⁴ Solomon Ells for the EU weighted average and EU best practice are 80.5 and 59 respectively. This is equivalent to an energy efficiency improvement potential of 27% in the EU petroleum refineries (Schyns, 2006).

FIGURE 7:



Solomon benchmark curves and the estimated benchmark curve for steam crackers benchmark, 2005

Note: the black line shows the regional current average SEC with respect to cumulative ethylene production (*primary* x-axis); the colored lines show the energy use of individual plants (*secondary* x-axis denoted by study participation (expressed in %)). SECs are expressed in terms of energy use per tonne of HVC.

Source: OGJ, 2003; IEA, 2003a, b; Solomon, 2005 in Leucx, 2008; own estimates

TABLE 4:

Energy use of the steam cracking process (in GJ/t HVC) and the improvement potentials

	2006	International and regional benchmark (2005)	Improvement potentials (%)	
Global1	16.9	12.5	25	
Europe1	~15.6	13.7	20	
North America	~18.3	15.8	32	
Asia-Pacific	~12.6	11.2	1	
China	~17.1	-	27	
India	~17.1	-	27	
Brazil	~18.3		32	

Source: Saygin et al., 2009 for year 2006; Solomon, 2005 in Leucx, 2008.

Note: Improvement potentials are estimated by comparing average 2006 values with the *international benchmark* (estimated at the 1st decile in the global benchmark curve as 12.5 GJ per tonne of HVC).

1 The 2001, 2003 and 2005 surveys covered 70% (14 Mt), 89% (21 Mt) and 71% (17 Mt) of the total European production respectively. The worldwide participation rates in the same years expressed in terms of physical production were 39 Mt, 69 Mt and 66 Mt respectively (Leucx, 2008). In 2005, the participation rate was equivalent to 50% of the total global production that year. In the absence of reliable production statistics covering earlier years, worldwide participation rates cannot be estimated.

TABLE 5:

IFA benchmark survey results for 2004 and 2007 (lower heating value, in GJ/t NH₃)

	2004 (plant)	2004 (production)	2007 (plant)	2007 EU-27 (plant)
Lowest SEC	28.0	28.0	23.5	27
1st decile	30.3	29.7	31.5	28.7
Last decile	43.5	43.3	43.0	42.9
Highest SEC	53.2	53.2	58.0	46.7
Average		36.9	36.6	35.7

Note: Plant: results of benchmark survey results based on individual plants; Production: results of benchmark survey results based on production capacity.

2004 Benchmark survey (production) is based on Lako, 2009. In the study, SEC data for 66 plants was distinguished between three categories on the basis of plant capacity, namely (1000 mtpd, 1000-1500 mtpd and >1500 mtpd. In order to rank 66 plants with respect to same production scale, we assume that plants utilised 100% of their reported capacity.

Source: 2004 Benchmark (66 plants): PSI in EFMA 2008; 2004 Benchmark (production):PSI in Lako, 2009; 2007 Benchmark (93 plants): IFA in Gielen, 2009; 2007 Benchmark EU-27 (35 plants): IFA in Ecofys, 2009.

FIGURE 8:





Note: Black and red lines refer to cumulative production (*primary* x-axis); all other curves refer to the plant number (*secondary* x-axis).

Source: 2004 Benchmark (66 plants): PSI in EFMA 2008; 2004 Benchmark (production): PSI in Lako, 2009; 2007 Benchmark (93 plants): IFA; 2007 in Gielen, 2009 Benchmark EU-27 (35 plants): IFA in Ecofys, 2009; USGS, 2009a; own estimates.

Ammonia

Figure 8 shows five benchmark curves, four of which were prepared by Plant Survey International (PSI). The fifth is based on new analysis for this study. Two of the curves prepared by PSI refer to the year 2004, one referring to cumulative worldwide production (red line) and the other to specific plants (yellow line). The third (blue) and fourth (green) curves prepared by PSI refer to the year 2007 and represent ammonia production worldwide and in the 27 EU member States (EU-27) respectively. Our own estimates (black line) depict the sector's energy efficiency on a regional basis. This is based on SEC values for ammonia production in different regions from various sources (IEA 2007; 2009d; Schyns, 2006; Nand, 2008; NRCAN, 2008; Yara, 2008; Lako, 2009; Zwiers, 2009; and from papers presented at IFA technical conferences). These regional datasets refer to the sector's energy use in years between 2002 and 2006, corrected to the reference year 2007 by the

application of an estimated energy efficiency improvement of 0.5% a year.

We determine the international benchmark as the ammonia plant at the 1st decile on the 2007 IFA benchmark curve (*i.e.* 31.5 GJ per tonne of ammonia). Since the most efficient plants are operated on natural gas, we assume that all plants in the world will switch to natural gas in their processes.

Methanol

A detailed plant inventory of the global methanol capacity to the end of 2006 has been conducted by the Methanol Institute (MI, 2009). Ideally, this inventory should for each country or region produce information on capacity, feedstock type, the first year of operation (or age) and the technology applied in each individual plant. Much of this data is absent for China, Brazil and Russia²⁵. The analysis therefore excludes the entire capacity in China and parts of the capacity in Brazil and Russia.

FIGURE 9:





Note: Production is estimated by multiplying the total capacity with a capacity utilisation rate of 85%.

²⁵ These countries account for more than a quarter of the installed methanol capacity worldwide. China's capacity is unknown, but is estimated to amount to more than 15% of the total worldwide capacity.

The largest plants are operated in South America (particularly in Trinidad and Tobago and Chile) and in the Middle East (in Iran and in Saudi Arabia). These two regions accounted for more than half of the global capacity in 2006. On average, Europe and some new plants installed in these developing countries are the most energy efficient regions in methanol production with an estimated average SEC, including feedstocks, of approximately 33 GJ per tonne of methanol (Figure 9). The plants with the highest energy use are operated in developing Asia including India and in transition economies. The energy efficiency improvement potentials compared to the international benchmark are approximately 10% to 15% in

Europe and as high as 25% in economies in transition and in India.

c. Non-ferrous metals

The non-ferrous metal industry is responsible for the production of aluminium, copper, chromium, nickel, zinc and other non-ferrous metals. The aluminium industry has the most detailed SEC and production data available, collected by the International Aluminium Institute (IAI). The analysis in this study reviews the production of metallurgical alumina, the raw material for primary aluminium, and the production of primary aluminium in smelters. Using benchmark surveys, copper smelters and slab zinc production are also analysed.



FIGURE 10: IAI benchmark curve and estimated benchmark curve for alumina production, 2007

Black line: average SEC of world regions. Red line: Plant specific SEC, excluding China and parts of several other regions (IAI, 2009).

Sources: IAI, 2009a,b; IEA, 2009c; Liu et al., 2009; Xiao-wu et al., 2009 own estimations

Alumina production

Based on IAI statistics, Figure 10 shows two sectoral energy use graphs for alumina production. The black line shows new estimates which include China and other countries missing in the IAI statistics. The red line shows the global benchmark curve prepared by the IAI, which covers 60% of world wide alumina production. Compared to the international benchmark (7.8 GJ/tonne, based on the global benchmark curve) North America and East Asia (including Oceania), which account for 60% of global production, have the potential to improve performance by only a few percent. Europe (including Central Europe and EIT) and Africa have the potential to improve performance by around 35% and China by up to 50%.

Primary aluminium production

Smelting is the most energy intensive step in the production of primary aluminium. The vast bulk of the energy used is used in the form of electricity. Figure 11 shows two benchmark curves for aluminium smelters. The black line shows new estimates which include China and other regions. The data originate from IAI statistics except for China for which the data are derived from on other literature. The red line shows the extended global benchmark curve which also includes China. The plant SEC data is collected by IAI. Excluding China, it covers 65% of world production (IAI, 2009a). We have estimated the energy use of the Chinese sector separately based on publicly available data (Zunhua, 2008; Yanjia and Chandler, 2010), which raises the production coverage to approximately 95%.

FIGURE 11:





Notes: IAI data refer to the SEC in primary aluminium production used for electrolysis by the Hall-Heroult processes (including rectification from alternating current to direct current). The data include smelter auxiliaries (including pollution control equipment). They exclude the power used in casting and in carbon plants. Bosnia and Herzegovina, Iran, Poland and Romania, in total representing less than 2% of the total global primary aluminium production, are excluded from this dataset.

Source: IAI, 2009a,b; Zunhua, 2008; own estimates

The region with the lowest SEC is Africa. This is possibly a result of recent investments in modern, large-scale plants in Mozambique and South Africa (BHP Billiton, 2006). On average, plants in China (which account for 77% of Asia's production capacity, under which they are reported) are also relatively efficient.

Worldwide, the average energy use is 15 560 kilowatt hours (kWh) per tonne of primary aluminium. The IAI is seeking to reduce the average to 14,500 kWh per tonne of primary aluminium, either by revamping or by replacing existing smelters (IEA, 2009c). This would achieve electricity savings of around 7%. New plants have energy use as low as 13,500 kWh per tonne, suggesting that up to 13% of current smelter electricity use could be saved. According to the IAI benchmark survey, the electricity use of the international benchmark is 14 215 kWh per tonne of primary aluminium produced²⁶.

Copper smelting

Sulphide ore concentrates, currently accounting for approximately 80% of the total primary copper production, are smelted and then refined to obtain high purity copper (Ullmann's, 2007). On the basis of the copper smelters benchmark survey (**Figure 12**), the energy use of the international benchmark is estimated to be 7.4 GJ per tonne of copper. The average energy use worldwide is 13.8 GJ per tonne. This implies the existence of a global improvement potential of 46% in the energy efficiency of copper smelters.

FIGURE 12:





Note: Data is provided in Partinen (2008) as total net energy consumption versus total copper production for each copper smelter.

Source: Data is reproduced based on Partinen, 2008

²⁶ The international benchmark is estimated by including China with its estimated average SEC value in the IAI survey. Given China's large volume of production, this increases uncertainty significantly.

²⁷ Brook Hunt is a research and consulting company which benchmarks the production energy use and production costs of nonferrous metals (e.g. copper, zinc, etc).



FIGURE 13: Brook Hunt benchmark curve for zinc smelters, 2006

Source: Data is reproduced based on Kouw, 2009

Slab zinc production

Slab zinc is produced by smelting zinc ore concentrates either through a pyrometallurgical process or through electrolysis. Using a benchmark survey for zinc smelters (**Figure 13**), the international benchmark energy use is estimated to be 15.2 GJ per tonne of zinc, based on the performance of smelters in Europe. The average SEC worldwide is 23.6 GJ per tonne. This indicates a global improvement potential of 36%.

d. Iron and steel

The primary output of the iron and steel sector is crude steel. Across the world, four major routes are applied for the production of crude steel:

- Blast furnace (BF)/Basic oxygen furnace (BOF)
- Smelt reduction/Basic oxygen furnace (BOF)
- Direct reduced iron (DRI)/Electric arc furnace (EAF)

Scrap/Electric arc furnace (EAF)

The process shares of crude steel production differ between countries. The most commonly used processes are BOF and EAF. BOF accounts for approximately two thirds of worldwide production, and EAF for slightly less than onethird. Around 3% of the worldwide capacity is based on open hearth furnaces. These are being phased out.

For each country, an EEI value is estimated which reflects the process mix and includes the production processes of the most important endproducts into which crude steel is further processed²⁸. These are:

- Hot-rolled flat products,
- Hot-rolled bars and concrete reinforcing bars, and
- Wire rod.

²⁸ The final energy use of the iron and steel sector is only partly reported in the IEA Energy Statistics under iron and steel. Although the total final energy use for the production of crude steel from pig iron and the power use in blast furnaces are included under this item, the fossil-fuel requirements of the blast furnace are excluded. The total energy consumption for these two items is added together.

Plants in Asia/Pacific operate with the lowest energy use (EEI=1.15), followed by the plants in Europe (1.3) and North America (1.35). Iron and steel plants in India (1.55), Africa (1.8) and EIT (2.25) have relatively high levels of energy use. EEI values can also be expressed in terms of potential energy savings (in GJ) per tonne of crude steel production compared to BPT. Regional averages on this basis are shown in Figure 14 as purple dots with respect to the secondary y-axis. The savings potentials per tonne of crude steel do not necessarily follow the same ranking as the EEI. This reflects structural differences in the activities of the sector between countries. In countries with a high EEI and a high share of secondary steel production, the specific improvement potentials tend to be lower than in those countries that have an equally high EEI but produce more primary steel. This can be seen, for example, in the figures for North America and for Europe.

The Asia Pacific Partnership (APP) is increasingly active in collecting comparable and consistent data on the energy performance of BOFs and EAFs operating in Australia, Canada, China, India, Japan, Korea and the United States. Together, these account for around 60% of the total global iron and steel production. The APP's latest industry survey shows that the most efficient BOF in the region has an energy use of 18.2 GJ per tonne of steel and the least efficient blast furnace uses 40.9 GJ per tonne of steel. The best EAF has a specific energy use of 6.2 GJ per tonne of steel and the least efficient EAF uses 30.1 GJ per tonne of steel (APP, 2008). The coverage of the study is complete for Australia, Canada and Korea and partly so for Japan with limited coverage of EAFs. The data for all other countries have major gaps. Given this patchy coverage, the APP results are not used in the present analysis.





Note: The benchmark curve is based on the left hand y-axis. The dots show the specific improvement potentials in each region relative to BPT based on the right hand y-axis.

Source: WSA, 2009; IEA, 2006a, b

e. Non-metallic minerals industry

The non-metallic minerals sector includes the production of cement, lime, bricks, glass and ceramics. In most countries, cement and lime production are by far the largest energy users in the sector, accounting for more than 80% of the sector's reported energy use according to international energy statistics. Among the subsectors, the cement sector has the best developed arrangements for the collection of data on energy and CO₂ emissions, through the Cement Sustainability Initiative (CSI). The glass industry has made efforts to produce similar data. But the production coverage is limited and the publicly available results are now out-dated. For the lime sector, a recent paper by EuLA (2009) provides benchmark curves for the European lime sector for horizontal and vertical kilns. But no data yet exist on which a global benchmark could be based.

FIGURE 15:





Black line: own estimates based on CSI (2009a) and IEA (2009c). Red line: based on the GNR database.

Note: "China GNR" is estimated based on GNR database which covers 4% of China's cement production (from a total of 60 plants). "China Other" is our own estimate for the remaining 96% of production, based on a total China average of 4.1 GJ of thermal energy per tonne of clinker (IEA, 2009c).

Source: GNR Database (CSI, 2009a); own estimates.

Cement

Clinker plants with the lowest final energy use²⁹ are operated in India at 3.1 GJ per tonne of clinker followed by the plants in the Pacific region and other developing Asia at around 3.3 GJ per tonne (**Figure 15**). The least energy efficient plants are located in North America at 4.2 GJ per tonne and in EIT countries at 6 GJ per tonne. The energy use of the BPT (a six-stage pre-heater and pre-calciner kiln) is 2.9 GJ to 3.3

GJ per tonne of clinker. The benchmark curve developed for this study indicates that the worldwide average energy use for clinker production is approximately 3.9 GJ per tonne. On this basis, the average energy saving potential is around 0.6 GJ to 1 GJ per tonne of clinker, *i.e.* approximately 24%.

The electricity consumed in cement plants (for grinding) is surveyed separately by GNR (**Figure 16**).

FIGURE 16:





Black line: own estimates based on CSI (2009a) and IEA (2009c). Red line: based on GNR database.

Note: "China GNR" is estimated based on GNR database which covers 4% of China's cement production. "China Other" is our own estimate based on 115 kWh grinding electricity use per tonne of cement (IEA, 2007).

Source: GNR Database (CSI, 2009a); own estimates.

²⁹ Total final energy use for cement production in each country is estimated as the sum of (i) fuel SEC of clinker production multiplied by clinker-to-cement ratio and (ii) electricity SEC for cement production.

TABLE 6:

Specific energy consumption of lime kilns in selected countries

	EU	J-271						
Region	Heat use (GJ/t)	Kiln electricity use (kWh/t)	US	Canada	China	India	Thailand	
Horizontal kilns								
Long rotary kilns	6-9.2	18-25		7-13	-	-	-	
Rotary kilns with preheater	5.1-7.8	17-45		6-9	-	-	-	
Vertical kilns	-	-	-	5-7	4.5		<13.2	
Parallel flow regenerative kilns	3.2-4.2	20-40	-	-	-	-		
Annular shaft kilns	3.3-4.9	18-35 (50)	-	4-4-5	-	-	-	
Mixed feed shaft kilns	3.4-4.7	5-15	-	-	-	-	-	
Other kilns	3.5-7.0	20-40	7.23	7.2	-	5.6	-	

¹ Lower bound: large-scale kilns; upper bound: small-scale kilns.

² Data refer to small vertical kilns (Dankers, 1995).

³Refers to the energy use of lime kilns operated in US pulp and paper mills (Miner and Upton, 2002).

Source: NRCAN, 2001; CIEEDAC, 2004; IPTS, 2010; Venkatarama Reddy and Jagadish, 2003; Wei, 2007; IEA, 2007; Dankers, 1995; Miner and Upton, 2002

Lime production

Global lime production including captive lime³⁰ is 172 Mt (IPTS, 2010). More than 40% is produced in China, and 16% in Europe (EU-27). In Europe (including EFTA and Croatia and Turkey) there are around 600 kilns producing lime other than for captive uses. In 2006, 5 000 kilns in China produced 75 million tonnes of lime. Most of these plants are small-sized and approximately 60% of them are based on outdated earthen kiln technology. A switch to the use of semimechanised and mechanized vertical kilns would achieve energy reductions of around 20%. In addition, almost all the lime kilns in China are fuelled by anthracite coal.

Glass production

Glass production typically involves the mixing of raw materials, melting in furnaces, forming

and post-processing (including annealing and finishing). The first two processes are identical regardless of the type of final glass product. Melting accounts for the largest share of energy use in a glass plant (~80%). A 1999 benchmark survey covering 123 container glass and 23 flat glass furnaces operated worldwide found a difference of approximately a factor two between the best furnaces operating at 3 850 kJ/kg of glass produced, and the least efficient ones operating at more than 8 ooo kJ/kg of glass (Beerkens and Limpt, 2001). The average energy use was 5 200 kJ/kg of glass (Figure 17). Generally, float glass production consumes more energy than container glass production. The SEC for specialty glass products such as TV panel glass is even higher, due to higher quality requirements (Ullmann's, 2007), but no reliable data is available at country level.

³⁰ Captive lime is lime produced for internal consumption in integrated plants (in sugar, pulp and steel industries).



FIGURE 17:

Energy requirements of 123 continuous container glass furnaces, 1999 (normalized for 50% cullet)

Source: Data and figure are reproduced based on/from Banarjee, 2006; Sardeshpende et al., 2007

TABLE 7:

Overview of specific energy consumption in continuous glass furnaces (in kJ/kg of melted glass)

	Average	1 st dec	ile (or BPT)		Production	
	Container	Flat	Container	Flat	volume (M\t/yr)	
Global ¹	~5,200	-	~3,850	-	~130	
US	4,000-10,000 (6,065)	5,000-8,500 (6,860)	3,270	3,690	~21	
EU-27	4,000-10,000 (~5,000²)	5,800-8,700	4,2 00 ³	6, 300⁴	~33	
China⁵		7,800		-	~27	
India ⁶	6,800	-		~4,400	~2	
Russia				-	~5.2	

Note: Production volumes are for 2005 and include the production of container glass, flat glass, glass fibres (incl. mineral wool, textile and optical) and other glass products, *e.g.* specialty glasses.

¹ Data refer to **Figure 17** where the benchmark curve represents the situation in 1999 only. Data for US and EU-27 refer to more recent years. Other studies report much lower energy use at approximately 2 650 kJ/kg for BAT (Kobayashi *et al.*, 2005). To apply a consistent methodology across all countries, we use the results of the 1999 benchmark survey.

² Data in brackets are own estimate based on the survey results of 256 furnaces. Data is corrected for 50% cullet. ³ The most efficient plant applies end-fired regenerative technology. If energy required for oxygen production is excluded, then oxy-fuel technology represents the BPT with a SEC of 3050 kJ/kg to 3500 kJ/kg melted glass. Data is corrected for 50% cullet.

⁴ Technology refers to modern energy-efficient cross-fired glass furnace with regenerative air preheating (float glass). Data is corrected for 20% cullet.

⁵ The survey includes 28 float glass production lines.

⁶ The survey includes 17 furnaces.

Source: Beerkens and Limpt, 2001; IPTS, 2009draft; Banarjee, 2006; Rue et al., 2007; Wang, 2007; Sardeshpende et al., 2007; Sarkisov et al., 2007

Table 7 compares the SEC of continuous glass furnaces operated in the four major manufacturing regions, the United States, EU, China and India. On average, US glass furnaces have the lowest SEC for container and for flat glass production. Furnaces in Europe have higher energy use for container glass production. Glass furnaces in China have the highest energy use. The energy used by batch furnaces is much higher than that used by continuous glass furnaces, ranging from 12.5 GJ to 30 GJ per tonne of glass produced (Römpp, 1995). Worldwide, the potential energy saving from moving all plants to the efficiency of the BPT amounts to approximately half of the sector's current energy use.

Ceramics

Bricks and tiles, fine ceramics, sanitary stoneware, and similar products are produced by the ceramics sector. In this section, we elaborate on brick making only due to significance of its energy use in SMEs and small-scale industries (GEA, *in preparation*) and to the availability of numerous data in literature. There are currently 300 000 kilns operating worldwide. Four countries, China (54%), India (11%), Pakistan (8%) and Bangladesh (4%) account for approximately 75% of the worldwide production (Chaisson, 2008).

TABLE 8:

Specific e	nergy	consumption	of	different	brick	making	technologies	; in	selected	countries	(in	GJ/	/t
------------	-------	-------------	----	-----------	-------	--------	--------------	------	----------	-----------	-----	-----	----

	Intermittent kilns	Bull's trench kilns	Hoffmann kilns	Tunnel kilns	VSBK kilns
Bangladesh ¹	2-4.5	1.152			
China	2.5		1.2-1.5 ³	1.3-1.5	
India	3-11 ⁴	1.8-4.2 1.1-1.8	1.5-4.3	1.5-2	0.7-1.0
Indonesia¹	2-4.5				
Nepal ¹	2-4.5				0.7-1.0
Sri Lanka	5-65				
Thailand ¹	2-4.5				
Vietnam	2.2-3.1			1.4-1.6	0.7-1.0
Brazil	2-3				
Bolivia			1.5-6		
Egypt			9-11		
Europe				1.5-3 ⁶	
BPT	1.5 (clamp)	1.9 (0.75) ⁷	1.5	1.5	0.75

Note: 1 brick weighs 2.5 - 3 kg.

¹ Data given for intermittent kilns is average for Asia (Heierli and Maithel, 2008).

² Fixed chimney kilns.

³Artificial drying adds another 0.1-0.25 GJ/t energy use.

⁴Scotch kilns have an energy use of 1.5 GJ - 7 GJ/t. Downdraught kilns may use 9.3 GJ/t.

⁵ Refers to the performance of downdraft kiln where drying takes place via natural draft.

⁶Refers to firing and drying in tunnel kilns. The manufacture of clay blocks requires slightly lower energy. In 2003, average SEC in the EU was 2.3 GJ/t (IPTS, 2007). A modern tunnel kiln in Germany uses 1.1 - 2.5 GJ/t (Heierli and Maithel, 2008).

⁷ Value in brackets refers to the best fixed-chimney Bull trench kilns.

Source: AIT, 2003; Zaiyin, 2007; Schwob et al., 2009; UNFCCC, 2006; Lebbing, 1999; Nurhayati et al., 2006; IPTS, 2007; Heierli and Maithel, 2008; ESMAP, 2007; FAO, 1993

China is the largest brick producer worldwide, operating around 90 000 kilns for brick production with an output of around 900 billion bricks, two-thirds of which are solid burnt clay brick and the rest new wall materials (Zaiyin, 2007; Chaisson, 2008). 90% of the production in China is based on Hoffmann (annular) kilns. India is the second largest brick producer worldwide with an output of around 140 billion bricks from 100 000 kilns of which 70% are Bull trench kilns (Maithel, 2002; Chaisson, 2008). 80% of the capacity in South Africa is based on clamp kilns. Three quarters of the production in Bangladesh is based on more than 3 000 fixed chimney kilns (Ferdausi *et al.*, 2008).

Several kiln types are used in brick making. The most energy intensive kilns are intermittent kilns, such as clamp, scove and Scotch kilns, where bricks are fired in batches (Table 8). Continuous kilns are more energy efficient due to continuous firing. They come in two main types, depending on whether the bricks or the fire move during the process. The bricks move in tunnel kilns and vertical shaft brick kilns (VSBK) developed in China. The fire moves in Hoffmann kilns and Bull trench kilns.

VSBK is the most energy efficient kiln type **(Table 8)** and is the technology of choice in several South East Asian countries such as India, Nepal, Pakistan and Vietnam. VSBK kilns require a relatively high capital investment³¹.

f. Pulp and paper sector

The main activities of the pulp and paper sector are chemical pulping, mechanical pulping, paper production and paper recycling. Unlike any other industry sector, the pulp and paper sector is a large generator of energy, particularly in kraft (or chemical or sulphate) pulp mills, in the form of black liquor. Black liquor is converted into heat and power which is consumed internally either in the pulp mill itself and/or at the paper mill if the site is an integrated one. Currently, around half of the sector's energy demand is met by biomass residues in this way. Mechanical pulping uses energy in the form of power only.

Energy integration in the pulp and paper sector makes it difficult to account consistently for the sector's energy demand. While many countries produce chemical pulp, the biomass energy used is not reported under the pulp and paper sector in the international energy statistics, but instead under other non-specific industries (IEA, 2009c). This makes the development of an EEI for the pulp and paper sector based on energy statistics very uncertain (**Figure**). Some regions such as developing Asia, Asia/Pacific and Europe have EEI values less than 1. This implies that they are more efficient on average than BPT. This is not technically possible. Other regions have EEIs as high as 2 to 3.

The sector's energy efficiency potentials compared to BPT as reported by the IEA are shown in Table 9 (IEA, 2009c). The largest potentials exist in plants operated in Russia, followed by plants located in the United States and Canada, reflecting the continuing use of older capital stock in these regions. The capacity in Europe, particularly in Finland (Jokinen, 2006; IEA, 2009c) and in Brazil is amongst the newest. As a result, these regions have only limited further energy efficiency potentials. Chinese plant has a large potential to improve energy efficiency (IEA, 2007). Estimates of low specific energy saving potentials in China underline the degree of uncertainty in the energy statistics.

³¹ VSBKs are generally used to fire solid bricks in South Asia although they are used extensively to fire hollow bricks in Vietnam. According to literature, this technology can fire bricks with 15-20% hollows only; for larger hollows high breakage rates are observed. Therefore VSBK technology has limitations in its ability to fire a large variety of clays and its suitability to fire a wide range of clay products (Heierli and Maithel, 2008).



FIGURE 18:

Estimated benchmark curve for the pulp and paper industry, 2006

Source: FAOSTAT, 2009a; IEA, 2009a, b, c

TABLE 9:

Theoretical energy saving potentials in the pulp and paper sector compared to BAT, 2006

Region	Specific improvement potential (GJ/t)
OECD North America	5.2-7.0
OECD Europe	0.6-2.0
OECD Asia	0.2-0.5
Brazil	2.4
China	0.9
Russia	11.6

Source: IEA, 2009c

g. Foundries

The foundry sector produces cast metal products based on either ferrous or non-ferrous alloys. Although there are reliable statistics for a number of foundries and regional physical production volumes as well as for region-wide SEC values, the foundry sector is not separately reported in international energy statistics. Foundries are dispersed across a number of sectors against which their activities are accounted for, such as the iron and steel, non-ferrous metals, and machinery sectors.

TABLE 10:

Foundry benchmark electricity use (in kWh/t of melted product, furnace electricity consumption)

	EU	Range Canada	US¹	India ²	International Benchmark
Cast iron energy consumption	520-800	595-1,290		780- 900	520
Alloy cast steel	500-800	620-2,760		735	500
			500-825		
	1,360	400-1,100			
	for Cu	for Cu		590	400 for Cu
Non-ferrous	600-1,250 for Al	570-1,610 for Al			570 for Al

Note: As there is no global benchmark, improvement potentials are estimated compared to the region with the lowest SEC value.

¹ Data refers to electricity use in melting furnaces regardless of the type of metal melted. Lower energy use refers to modern and efficient induction furnaces. Higher energy use refers to electric-resistance heated reverbs. While the lower range for US is slightly less than the lower range given for EU, as it is not clear which type of metal melt this dataset represents it is excluded from the analysis.

² Data refer to the energy use performance of a single company in India.

Source: EU: IPTS, 2005; Canada: NRCAN, 2003; US: Energetics, 1999; India: Kirloskar, 2009

A survey of Canadian foundries covering a total of 45 foundries found that the melting process in iron foundries accounts for 54% - 84% (average 66%) of the total final energy use (NRCAN, 2003). The remaining energy use results from electricity consumption by motors (30%) and lighting (4%). 12% is consumed in air compression systems which account for around 100 kWh to 200 kWh per tonne of cast product. for example in sand casting³². Although lighting demand is very similar in steel and bronze and copper foundries, the energy demand for motors is higher at around 57% in bronze and copper foundries than in steel foundries (around 47%). The share of electricity demand for the melting process is however higher in steel foundries at around 45% - 65% (average 49%) than in bronze and copper foundries at around 38%.

h. Textiles

The production of textiles involves spinning, weaving, knitting and wet processing (including dyeing, printing and finishing). Most of the sector's output and energy use is located in developing countries. There are large data gaps in the energy statistics, production data and SEC values for these countries. The analysis therefore focuses on spinning and weaving, the processes for which most information is available.

Spinning

The output of the spinning process is yarn. Yarn is produced either from natural fibres such as cotton, man-made fibres such as polyamide, or a blend of these two fibre types. Total yarn production in 2007 was 63.5 million tonnes.

³² Sand casting is a type of process applied in the production of products weighing at most 100 kg per piece. Electricity consumption for compressors is similar in the Indian foundry sector, 100-150 kWh/t in average sized foundries and 50-75 kWh/t in smaller foundries (Gandhe, 2009).

There are several types of spinning technology. Two of them, ring spindles and open-end spinning (OE), dominate. The oldest spinning technology, ring spinning, dominates the installed capacity because of the high quality yarns it produces. However, compared to other spinning systems, this technology has lower production speed and higher energy consumption (Oxenham, 2002).

Table 11 shows SEC values for modernised ringyarn and OE spindles for number of countries,based on data from a survey by the InternationalTextile Manufacturers Federation (ITMF) alongsidedata based on various studies for other importantmanufacturing countries. Energy consumption

depends both on the technology used and on the yarn count (expressed in *e.g.* tex, Ne)³³ as it determines the total production per spindle (Spinnovation, 2007). The end-use purpose of the yarn influences the final energy demand. For example, energy use increases by 20% for yarns suitable for weaving compared to knitting yarns (Koc and Kaplan, 2007). The type of fibre used has no significant effect on energy use.

Weaving

In the weaving process, yarns are processed in looms to produce woven fabric or cloth. Generally two types of looms are used in weaving, handlooms and power looms (the latter either with shuttle or shuttles).

	Ring yarn (combed)	SEC (kWh/kg) Open end yarn	Range (for all technologies)
Brazil	3.54	2.58	
China	3.49	2.58	
India	3.57	2.5	1.9-61
Thailand ²			0.55-7.3
Indonesia			1.87-5.04
Italy	3.52	2.57	
Korea	3.62	2.55	
Turkey	3.56	2.44	< 4 ³
USA	3.57	2.57	

Specific energy consumption values of spinning technologies in different countries

TABLE 11:

Note:	Although	no i	nternational	benchmark is	available,	we	estimate	improvement	potentials	compared	to	the
region	with the	e low	est SEC.									

1 SEC for cotton yarn production is slightly higher than 4 kWh/kg; the SEC for general yarn is less than 6 kWh/kg.

2 For Thailand, additional fuel consumption of 0.14-0.73 GJ/t is reported.

3 Approximately 20% of the total power used is consumed in air-conditioning.

Sources for specific technologies: ITMF data based on Koc and Kaplan, 2007 Sources for ranges: India: Ray and Reddy, 2008; Indonesia: PREGA, 2005; Thailand: Visvanathan et al., 2000; Turkey: Turna, 2002

³³ According to a recent study, most short staple yarn production is between yarn count Ne 18 and Ne 32. Yarn count to a maximum of Ne 60 covers 90% of the total short staple yarn production (Gherzi, 2009).

Across different weaving technologies, weft insertion systems consume a large share of the total electricity use of the equipment. Compared to the most efficient projectile looms, conventional shuttle and conventional rapier systems require much more energy in cloth production. Besides electricity use in the looms itself, there is also heat is also needed for preparation processes prior to weaving, for increasing yarn resistance and also for drying, if necessary **(Table 12)**. material processes and the large number of intermediate steps leading to final products.

Given the complexity of the sector, SEC values are provided only for a selected number of products which represent the bulk of the sector's energy use. These data are representative for Western Europe **(Table 13)**. Although the Statistics Department of the Food and Agriculture Organisation (FAOSTAT) provides production statistics for a number of basic food products for

TABLE 12:

Specific energy consumption values of weaving technologies in different countries

	Range for al		
	Electricity (kWh/kg)	Heat (GJ/tonne)	Total (GJ/tonne)
Germany			11-65
India	4.9-5.3		27-32.4
Indonesia	0.7-2	2.9-14.1	5.4-21.3
Thailand			5-43
Turkey	2.1-5.6	8.3-17	15.9-37.2

Source: India: Sathaye et al., 2005; Indonesia: PREGA, 2005; Ray and Reddy, 2008; TERI; Thailand: Rauch, 2009; Turkey: Turna, 2002

It is not possible to report either a global benchmark curve or EEIs for the textile sector or any of its products because most production is in developing countries for which there are no reliable relevant energy statistics. For example, for India, no power use is reported. And for Pakistan, the IEA reports the energy use of the textile sector under *other sectors* (IEA, 2009a, b).

i. Food and beverages

The food and beverage sector produces a wide range of intermediate and final food products. These products are produced by processing agricultural crops and from livestock. The industry consists of a number of sub-sectors, such as the dairy, meat, fruit and vegetables, grain mill products, beverage and other sectors. The food sector is very complex, given the wide variety of a large number of countries, no data are reported in international energy statistics for the food sector for most developing countries.

The products in **Table 13** cover between around 15% (for Brazil and Thailand) to 50% (for EU-27) of the sector's energy use³⁴. The coverage is too low to enable inter-country analyses of energy efficiency improvement potentials by developing EEI. For some developing countries, the data appear to account for more than 100% of the expected energy use in the sector. This suggests uncertainties in the reliability of the sector's reported energy data in international energy statistics. It is not appropriate to apply the OECD SEC data to analyse the global energy efficiency potential since operational conditions vary widely between ICs and DCs.

³⁴ If global production and SEC data were available, the production of other energy-intensive products of the sector could have been included in the analysis, e.g. tobacco, processing of fruits and vegetables (including preserving), other drinks and other food products such as flour, pasta, soups, sweets, etc. Other energy-intensive activities which are excluded are processing of starch products (especially corn and wheat), sugar and oil crops. These are important processes not only in ICs, but also in DCs, especially large producers of agricultural products (*e.g.*, Mexico, China, Brazil and South Africa (CRA, 2010).
TABLE 13:

Selected products and their specific energy consumption (in GJ of final energy per tonne of output unless otherwise stated) of the food and beverage sector in OECD countries

	Electricity	Heat	Total
Dairy sector			
Butter, ghee	0.5	1.3	1.8
Cheese	1.2	2.1	3.3
Fluid milk ^a	0.2	0.5	0.7
Milk powder	1.1	9.4	10.5
Condensed milk ²	0.3	1.9	2.2
Whey dry	1.1	9.9	11.0
Meat sector and fish production			
Dried, salted, smoked fish	0.01	2.1	2.1
Fresh, chilled, frozen fish	0.6	0.01	0.61
Fish meals	0.7	6.2	6.9
Carcass beef, veal, sheep ³	0.3	0.5	0.8
Carcass poultry ³	1	0.6	1.6
Carcass pork ³	0.5	0.9	1.4
Starch products			
Wheat starch	3	8.8	11.8
Maize starch	1	2.3	3.3
Potato starch	1.4	3.6	5
Other			
(Vegetable) Oil	0.2	2.7	2.9
Sugar (refined)	0.6	5.3	5.9
Cocoa beans ⁴	-	-	6.4
Coffee ⁵	0.52	2	2.5

¹ Fluid milk includes pasteurised, sterilised and long-life milk including the production of all fluid milk regardless of the fat content (whole, semi-skimmed or skimmed).

² Data include the production of both unsweetened (or evaporated) and sweetened condensed milk.

³ For meat products, FAOSTAT reports the meat yield per animal. We convert this to physical production volumes in tonnes by multiplying the data with the total number of animals slaughtered in each country. Data refers to per tonne of dress carcass weight.

⁴ For reasons of confidentiality, the original source provides the data in terms of primary energy (Neelis *et al.,* 2004).

⁵ Data refers to roasted coffee beans.

Source: IPTS, 2006; Neelis et al., 2004; Ramirez et al., 2006a; Ramirez et al., 2006b; NRCAN, 2005

Brewery

Benchmarking studies are available for the brewery sector (KWA, 2004), indicating that the average energy use per hectolitre (hl) of beer production has decreased by more than 2% per annum **(Table 14)**.

Detailed results from the 2003 brewery benchmark are shown in Figure 19 (KWA, 2004). This benchmark study covered 26% of the worldwide production. Most data originate from European breweries which together cover 43% of the total European beer production, followed by Australian and American breweries, with shares of 28% and 22% of their production respectively. The benchmark data suggests a worldwide potential to reduce the sector's energy use by approximately 30% to achieve the performance of plant in the 1st decile.

TABLE 14:

Results of the 1st, 2nd and 3rd brewery benchmark (years refer to the benchmark surveys conducted)

		Specific energy consumption			
Year	Number of breweries	Average	Standard Deviation	Median	Decile
1999	86	271	64	261	193
2003	158	239	60	233	176
2007	143	229	71	220	156

Source: Sharpe et al., 2009.





Source: Data is reproduced from KWA, 2004



UNITED NATIONS INDUSTRIAL DEVELOPMENT ORGANIZATION Vienna International Centre, PO. Box 300, 1400 Vienna, Austria Telephone: (+43-1) 26026-0, Fax: (+43-1) 26926-69 E-mail: unido@unido.org, Internet: www.unido.org